RISK AND RESILIENCE REPORT

Civil Protection Roadmap for Drought Resilience in Switzerland

Jurgena Kamberaj, Simon Aebi

Zürich, September 2024 Center for Security Studies (CSS), ETH Zürich





Authors: Jurgena Kamberaj, Simon Aebi

Contributor: Andrin Hauri

ETH-CSS project management: Simon Aebi, Co-Head of the Risk and Resilience Team; Daniel Möckli, Head of Think Tank Client: Federal Office for Civil Protection (FOCP) FOCP project supervision: Stefan Brem, Head of section "Future Development of Civil Protection"

Disclaimer: All views and opinions presented in this study are solely those of the authors.

DOI: 10.3929/ethz-b-000697719

Acknowledgements: The authors extend their gratitude to the cantonal participants of the survey and interviews, as well as the representatives from federal institutions for their invaluable contributions to this report. Their insights and expertise have been instrumental in fostering a deeper understanding of the multifaceted challenges and emerging opportunities in drought management within Switzerland.

Content

Execut	ive Summary	4	
Contex	ct	4	
Object	ive	4	
Metho	dology and Structure	4	
Results		6	
		-	
1	Introduction	7	
1.1	Definition and characterization of drought	7	
1.2	Drought in Switzerland	8	
1.3	The Swiss Civil Protection System		
1.4	Drought relevance for Swiss civil protection	9	
1.4.1	Drought-specific challenges for civil		
	protection	9	
1.4.2	Drought management capabilities in		
	the Swiss civil protection	11	
1.4.3	Swiss Civil Protection's evolution in		
	drought management	12	
	0		
2	Drought preparedness in Switzerland	13	
2.1	Survey results	13	
3	Drought mitigation toolkit:		
	10 Action Points	27	
	Action Point 1 – Establish and incorporate		
	a national drought early warning system	29	
	Action Point 2 – Assess drought risk	31	
	Action Point 3 – Include drought in the		
	cantonal hazard analysis	33	
	Action Point 4 – Translate drought risk		
	to local action	34	
	Action Point 5 – Develop a drought		
	response plan	36	
	Action Point 6 – Appoint a drought		
	task force	38	
	Action Point 7 – Strengthen drought		
	preparedness through exercises	40	
	Action Point 8 – Participate in knowledge		
	exchange	43	
	Action Point 9 – Conduct community		
	education and outreach programs	44	
	Action Point 10 – Embrace collaboration	46	
4	Concluding Remarks	48	
5	Bibliography	49	

Executive Summary

Context

Climate-exacerbated hazards, particularly hydrological extremes, increasingly affect human and environmental systems across Switzerland. With the population acutely aware of these events, numerous initiatives are underway to strengthen the preparedness of institutions, communities, individuals, and ecosystems.

Recent assessments, such as the ETH Center for Security Studies' Trend Analysis Civil Protection 2035 report,¹ indicate that climate change adaptation is recognized as a strength of the Swiss civil protection system. However, the recently published Capability Analysis for Civil Protection (*Fähigkeitsanalyse Bevölkerungsschutz*)² by the Federal Office for Civil Protection (FOCP) has identified deficiencies in prevention, prompting a more comprehensive examination of the system's capacity to address slow-onset and long-lasting crises like drought.

From a federal perspective, drought presents an opportunity to enhance Switzerland's overall crisis management capabilities. The currently ongoing improvement of the federal crisis management structure (*Krisenmanagement der Bundesverwaltung*)³ can be leveraged to enhance both short-term coordination and long-term advisory functions in response to drought events. Furthermore, the anticipated new security strategy from the Federal Council in 2025 may provide an additional avenue for addressing drought and other natural hazards as significant threats requiring comprehensive mitigation and adaptation strategies.

Objective

This report is part of continuing research initiated by the FOCP and focuses on the assessment of the civil protection capabilities needed to cope with climate-exacerbated hazards in Switzerland. This study specifically addresses the multifaceted challenges posed by drought to civil protection systems in Switzerland. It seeks to provide a roadmap that includes current practices, vulnerabilities, and future directions in drought mitigation within the Swiss context.

The primary objective is to strengthen the link between civil protection systems and drought mitigation by advocating for an integrated risk-oriented approach. To this end, the report assesses the current state of drought preparedness in Switzerland, with a particular emphasis on the evolving role of civil protection. In addition, the goal is to identify and map existing initiatives and practices, breaking down organizational silos and fostering a broader awareness and knowledge exchange.

Finally, this report aims to develop a practical and actionable Drought Mitigation Toolkit, which will serve as both a guiding resource for regional and cantonal preparedness plans and a foundation for future policy development in drought mitigation. The Mitigation Toolkit aligns with the integrated risk management cycle⁴ promoted by the FOCP, emphasizing early actions taken before or at the onset of drought to minimize its impacts. Additionally, the report advocates for the exchange of knowledge from past experiences to foster collaboration. In doing so, the report aims to strengthen Switzerland's overall resilience to drought and promote a more proactive, adaptable approach to managing drought-related risks.

The authors would like to express their gratitude to the cantonal civil protection authorities and national drought experts from the federal administration and federal institutions for their valuable contribution to this project. Their input has been essential to the development of this report, and we are deeply thankful for their collaboration and expertise.

Methodology and Structure

The research for this report employed a multidisciplinary approach that combined quantitative and qualitative research methods, including desk-based research, survey analysis, and expert interviews, as illustrated in **Figure 1**.

The initial phase of data collection began in April 2024 with the distribution of an online survey to authorities from all 26 cantonal offices of civil protection and civil defense in Switzerland. The survey was directed to those responsible for the strategic development of civil protection within their respective cantons, ideally heads of the Office of Military and Civil Protection/Civil Defense or members of the cantonal crisis management body (kantonale Führungsorgane or kantonale Führungsstäbe) involved in the cantonal hazard analysis (kantonale Gefährdungsanalyse) and preparedness planning (Leitfaden Kataplan).⁵ Survey participants were identified through research into each organizational structure of the cantonal Offices of Military and Civil Protection/Civil Defense. When this was not publicly accessible, personal networks were leveraged.

Grounded in the National Risk Analysis of Disaster and Emergencies (*Nationale Risikoanalyse "Katastrophen und Notlagen Schweiz"* (*KNS, English abbreviation: DES*)),⁶ which provides definitions and guidelines for drought management, the survey was structured in two sections. The first section collected baseline data on current drought management strategies, while the second evaluated responses to a hypothetical drought scenario, emphasizing strategic decision-making processes. The survey was administered via the platform Qualtrics in April 2024, with responses collected throughout April and May. Out of the 26 cantonal representatives invited, 25 accepted the invitation to take part. The responses have been aggregated to ensure the anonymity of participants and provide insights from the perspective of cantonal civil protection authorities, which may not necessarily reflect the views of cantonal governments or other agencies. A detailed overview of the survey results can be found in Section 2.1.

Subsequent to the survey, expert interviews conducted between April and May 2024 provided insights into drought mitigation efforts across Switzerland. The analysis of these interviews identified key themes, challenges, and best practices, directly informing the development of the Drought Mitigation Toolkit's 10 Action Points. While the detailed content of these interviews is not published, their influence is embedded within the Toolkit, alongside case studies and projects discovered through this process.

The first set of interviews was conducted with authorities from the civil protection and civil defense offices of six selected cantons – Basel-Stadt, Thurgau, Zurich, Bern, Fribourg, and Ticino – who are actively engaged in operational-level activities. This selection of the cantons ensured a diverse representation of Switzerland's geographic, linguistic, and regional landscape. The interviews provided an opportunity to exchange information, gather insights on past drought events, assess current challenges to drought mitigation, and understand existing measures and ongoing initiatives. To further complement the cantonal perspective, four additional expert interviews were conducted with national-level specialists involved in drought management across Switzerland. Representatives from the Federal Office of Meteorology and Climatology (Meteo-Swiss), the Federal Office for the Environment (FOEN), the Steering Committee on Intervention in Natural Hazards (LAINAT), and the Swiss Federal Institute for Forest, Snow, and Landscape Research (WSL) provided valuable insights into ongoing national efforts.

This report is organized into four sections. After this executive summary, which introduces the report's objectives, methodology, and an overview of the results, Section 1 presents a comprehensive overview of drought in Switzerland. Specifically, it provides a comprehensive overview of drought in Switzerland, starting with an examination of the phenomenon itself in Section 1.1 and its occurrence and impacts in Switzerland in Section 1.2. It then proceeds to describe the structure and functions of the Swiss civil protection system in Section 1.3 to prepare the ground for Section 1.4, which examines drought relevance for Swiss civil protection.

Section 2 presents the findings from the engagement with drought specialists and civil protection authorities. It begins by analyzing the survey results, aggregating responses from cantonal civil protection authorities in Section 2.1, and continues with insights gathered from interviews with national drought experts in Section 2.2.

Section 3 focuses on the Drought Mitigation Toolkit with its 10 Action Points for enhancing resilience

Target Group	Cantonal Strategic Civil Protection Authorities (n=25)	Cantonal Operational Civil Protection Authorities (n=6)	National Drought Experts (MeteoSwiss, FOEN, LAINAT, WSL) (n=4)
Methodology	Online Survey	Semi-structured Interviews	Semi-structured Interviews
Objective	 Collect baseline data on existing drought strate- gies Evaluate institutional preparedness to droughts 	 Gather detailed accounts of past drought events Understand current challenges Assess the effectiveness of existing measures 	 Explore drought phenomena in Switzerland Gain insights into national drought mitigation efforts

Figure 1 illustrates the methodology of the study

to drought. In conclusion, Section 4 summarizes the key findings and outlines the opportunities and lessons learned that can inform and strengthen Swiss civil protection efforts in preparing for and responding to complex crises like droughts.

Results

This report explores the significance of drought hazards for civil protection organizations in Switzerland, both at the local and national levels. The findings indicate that drought poses a substantial risk to civil protection systems across Switzerland's diverse geographical and regional landscapes. This significance arises from the impacts that droughts have on the population and their livelihoods as well as from the nature of civil protection systems, which includes crisis response and proactive measures such as prevention and mitigation.

However, the study also identifies several shortcomings, particularly in early action and drought prevention. These challenges were revealed through the survey of Swiss cantons and expert interviews, complementing the findings of the Capability Analysis for Civil Protection published by the FOCP.⁷ Particularly, the study points to key areas of concern:

- Fragmented drought management practices: Responses to drought are often uncoordinated among different cantons, leading to inconsistent water bans and bottlenecks in the water supply systems. The institutional framework for prevention and mitigation remains disjointed, although there is potential for improvement through the increasing integration of drought into cantonal hazard analyses.
- **Gaps in proactive planning:** While cantonal civil protection authorities are increasingly aware of the risks posed by droughts, proactive planning remains limited. Only a few cantons have established drought-specific management plans, while the majority focus on reactive crisis management, which includes short-term response measures.
- Challenges in water management: Key challenges in water management include inadequate infrastructure for water storage and distribution, difficulties in balancing water use between sectors such as agriculture, industry, and public supply, and a lack of effective enforcement of water use restrictions and conservation measures. These issues have, in the past, led to local conflicts among water users, highlighting the need for more cohesive and coordinated water management strategies.
- **Need for improved collaboration:** Collaboration in drought management across different administrative levels in Switzerland remains underdeveloped. The survey results show that while cantonal authorities

recognize the value of cross-border cooperation, few practical steps have been taken to formalize these relationships. Despite the limited collaboration, there is strong interest in participating in a national knowledge exchange program on drought management.

To address these shortcomings and to provide an overview of ongoing initiatives in Switzerland related to drought management from a civil protection perspective, the report identified 10 Action Points included in a Drought mitigation toolkit:

- Action Point 1 Establish and incorporate a national drought early warning system
- Action Point 2 Assess drought risk
- Action Point 3 Include drought in the cantonal hazard analysis
- Action Point 4 Translate drought risk to local action
- Action Point 5 Develop a drought response plan
- Action Point 6 Appoint a drought task force
- Action Point 7 Strengthen drought preparedness through exercises
- Action Point 8 Participate in knowledge exchange
- Action Point 9 Conduct community education and outreach programs
- Action Point 10 Embrace collaboration

These Action Points provide a roadmap for Swiss civil protection organizations to adapt and prepare for droughts. They acknowledge that drought is a nationwide concern, impacting regions to varying extents. The report also emphasizes the necessity for a shift in focus from reactive crisis management to proactive risk reduction.

While the report primarily targets the Swiss context, the identified Action Points and the underlying methodology can be adapted and implemented beyond Switzerland. They serve as a valuable resource for any region dealing with the challenges of drought, offering a practical framework to strengthen preparedness and mitigate the impact of this slow-onset hazard.

1 Introduction

1.1 Definition and characterization of drought

Drought is a complex phenomenon that is difficult to define and subsequently to monitor. In its simplest definition, drought is the absence of water caused by insufficient precipitation or excessive evapotranspiration linked to climate warming.⁸ When extended over a season or longer, this deficiency results in water supplies that are insufficient to meet the demands of human activities and the environment. Other factors, such as temperature, wind, and low humidity, can exacerbate a drought's severity and duration. To better understand its diverse impacts, drought is generally classified into:^{9, 10, 11}

- **Meteorological drought** is characterized by a lack of precipitation over a long period of time;
- **Hydrological drought** occurs when water levels in lakes and rivers drop below normal due to insufficient rainfall in their catchment area;

- Agricultural drought manifests when plant roots receive inadequate water, affecting crop yields and agricultural productivity;
- Socio-economic drought arises when water supply fails to meet societal and economic demand, leading to adverse consequences;
- **Ecological drought** occurs when water availability in the ecosystem drops below the threshold required to sustain natural ecosystems and their services to society.

Drought typically begins with a period of below-average precipitation, which can gradually propagate through the hydrological cycle. This can lead to reduced soil moisture, declining groundwater levels, deteriorating water quality, and decreased river flows. As drought intensifies, it can impact agriculture, resulting in crop failures and reduced yields. Further escalation can lead to water shortages for essential human needs like drinking water, industrial processes, and recreation. In extreme cases, drought can push ecosystems beyond a tipping point, causing irreversible damage and loss of biodiversity.^{12, 13}





Drought impacts are largely, drought impacts are largely non-structural and spatially pervasive, making them difficult to assess and respond to effectively. The lack of visually striking drought damage also poses challenges to media coverage and public awareness.¹⁴ However, the consequences of drought are severe, contributing to 650,000 deaths globally between 1970 and 2019. The economic toll is also significant, with annual global costs estimated at six to eight billion US dollars a year.¹⁵ In the United States, one of the few countries with available drought data, total losses attributed to droughts since 1980 are reported to be around 360 billion US dollars.¹⁶ These figures are likely conservative, as they often fail to capture the full spectrum of drought's indirect impacts, which are rarely systematically recorded by monitoring and reporting systems.

1.2 Drought in Switzerland

Although Switzerland boasts abundant water resources, with lakes and rivers comprising 4% of its territory and providing about 6% of Europe's drinking water,¹⁷ it is not immune to the growing threat of drought. An increase in drought occurrence has been observed for the period from 1981 to 2020, especially during the summer months.¹⁸ This is exacerbated by warming temperatures, causing precipitation to fall more often as rain than snow, leading to reduced snowpack in lower-lying areas.¹⁹ These changes have the potential to significantly impact Switzerland's water supply, particularly in remote communities that rely on snowpack, glaciers, and springs for water. The repercussions of drier, warmer conditions have also triggered ecological shifts, such as an increase in insect pests like the European spruce bark beetle (Ips typographus), posing a threat to forests that act as a natural protector from avalanches.²⁰

Moreover, the decentralized nature of the Swiss water supply network complicates the equitable distribution of resources during drought periods. Past drought events have sparked conflicts over water usage. For instance, prolonged dry periods have heightened the demand for agricultural irrigation, drawing from underground reserves that supply households.²¹ Additionally, rising water temperatures further exacerbate the situation, as the chemical, nuclear, financial, and technology sectors demand more water for cooling purposes.²² Tensions over water usage have also emerged along Switzerland's borders, with France seeking more water from Geneva's Seujet dam for its nuclear plants²³ and Italy demanding greater water flow through Lake Maggiore for irrigation purposes.²⁴

In 2022, Switzerland, like much of Europe, faced a historic drought. Europe experienced its hottest summer and second-warmest year on record, enduring its worst drought in 500 years, with far-reaching consequences for agriculture, forests, and glaciers.^{25, 26} Low water levels in the Rhine River caused significant delays in shipping, with some vessels operating at 25% of their cargo capacity.²⁷

Drought in 2022 affected as much as 15% of Switzerland's area. Comparably, the average annual area impacted during the years 2000–2020 was around 2.7%.²⁸ A 2022 cantonal survey²⁹ on the impacts of the drought revealed that many cantons experienced prolonged low water levels in small and medium-sized watercourses and groundwater levels hitting historic lows. Particularly regions south of the Alps were prone to drought, for example Lakes Maggiore and Lake Ceresio also reached recordlow levels. In addition, deteriorating water quality, as well as record-high water temperatures, led to fish mortality in some regions. Reduced discharge rates curtailed electricity production at hydroelectric power plants. In response, many municipalities issued water-saving appeals and restrictions on agricultural irrigation withdrawals became widespread. Conflicts arose locally among water users, with some reports of illegal water extraction.³⁰ In many cantons, short-term measures included the formation of crisis teams, the convening of working groups, and public communication efforts to mitigate the effects of the drought.

The summer of 2022 serves as a harbinger of a new climate in Switzerland,³¹ where droughts are expected to become more frequent and severe. The associated challenges could trigger cross-border or cross-sectoral crises hindering the sustainable and equitable use of Switzerland's water resources. As droughts increasingly threaten both lives and livelihoods in Switzerland, their management also falls within the scope of civil protection, whose federal mandate is "to protect the population and its livelihoods in the event of disasters [...] and to contribute to the limitation and management of damaging events".³² Swiss civil protection authorities across different administrative levels are tasked with making informed and inclusive decisions that integrate drought management into the broader disaster risk reduction framework.

1.3 The Swiss Civil Protection System

The Swiss civil protection system is a federally structured, multi-layered network consisting of five partner organizations: police, fire services, healthcare, technical services, and civil Defense (*Zivilschutz*). Each organization operates within its respective municipal, regional, and cantonal areas of responsibility, with the 26 cantons mandating and overseeing their activities based on the national law on civil protection and civil Defense,³³ as well as

their own cantonal legislation. This decentralized structure ensures localized responses tailored to the specific needs of the communities while allowing for flexibility in dealing with different emergencies. Complementing this localized approach, the FOCP plays a crucial role in assisting cantonal organizations in their civil protection mandate and responsibilities. In particular, it provides support and coordination for the cantonal hazard assessment and preparedness activities³⁴ as well as incidents that transcend cantonal boundaries, supporting a unified national response when necessary. Moreover, the FOCP is responsible for the national warning, alert, and information systems that notify authorities and the population of impending dangers and incidents.

Since the comprehensive revision of the Federal Law on Civil Protection and Civil Defense (Bevölkerungsund Zivilschutzgesetz) in 2019, the Swiss civil protection system has continued to evolve and adapt to emerging challenges. Several cantons have partially revised their legislation to incorporate these changes and to respond to recent events like the Covid-19 pandemic, the war in Ukraine, migration crises, drought events, and potential energy shortages. The Canton of Schaffhausen, for example, has adopted an integrated risk management approach that expands civil protection responsibilities from prevention to damage management and recovery, emphasizing the use of knowledge gained from past events to inform preventive measures.³⁵ The Canton of Bern has introduced a business continuity management project to enhance the cantonal authority's ability to function effectively in times of crisis.³⁶ In the Canton of Thurgau, revisions to the Civil Protection Act (Bevölkerungsschutzgesetz) now include all civil protection-related activities, not just those related to extraordinary events.³⁷ In the Canton of Basel-Stadt, the cantonal crisis organization is currently developing a resilience concept applicable to the cantonal institutions.³⁸ Other cantons, such as Valais and Aargau, have focused on strengthening coordination between local authorities and civil protection organizations through streamlined procedures and transparent information sharing.39,40

At the national level, the Swiss Federal Council has initiated a major restructuring of its crisis organization in response to lessons learned during the Covid-19 pandemic and recent world events.⁴¹ This restructuring aims to overcome siloed actions and thinking between Federal Departments and agencies as well as to promote a more integrated and coordinated approach to crisis management. Although still under revision at the time of writing, the renewed approach consists of three staff organizations: the Political-Strategic Staff (*Politisch-strate-gischer Krisenstab*), which will be responsible for developing options for action and preparing decisions for the Federal Council; the Operational Crisis Unit (*Operativer Krisenstab*), which will be tasked with compiling comprehensive information for decision-making; and the Permanent Core Staff (*Permanenter Kernstab*), which will ensure the FOCP and the Federal Chancellery are prepared for crises also during normal times.⁴² While these changes hold promise for improving crisis response at the national level, their integration with cantonal crisis management bodies, which have valuable experience in crisis management, remains an area for ongoing evaluation and potential refinement.

1.4 Drought relevance for Swiss civil protection

As drought hazards become increasingly relevant in Switzerland, their prevention and mitigation also fall within the scope of civil protection. This section explores the multifaceted aspects of drought management within the Swiss civil protection system, reviewing past achievements and future expectations. It begins by examining the conceptual challenges civil protection organizations face when dealing with slow-onset hazards like drought. The Section then delves into the identified deficiencies in the Swiss civil protection system's capacity to address drought. Finally, it outlines the ongoing shift within the Swiss civil protection system toward a more proactive and resilient approach to drought management, highlighting the challenges and opportunities that this transformation presents.

1.4.1 Drought-specific challenges for civil protection

In contrast to rapidly unfolding disasters, such as earthquakes, cyber incidents, or terrorist attacks, droughts develop gradually. They are often perceived as part of a natural cycle rather than a hazard with the potential to severely impact populations and their assets.⁴³ The slow onset of droughts, coupled with their expansive spatial and temporal nature, significantly complicates planning and readiness efforts.

Existing drought risk management plans, however, tend to prioritize crisis management and **response** over hazard **prevention** and **risk reduction**.⁴⁴ This is often due to a perceived lack of structural measures, such as water-efficient infrastructure or early warning systems, that could effectively prevent or **mitigate** droughts. Moreover, the tendency of relevant agencies to identify prevention measures that obviate the necessity for civil protection intervention altogether can lead to neglecting the importance of building resilience skills within the civil protection system.

The emphasis on crisis response also shapes how civil protection organizations distribute their resources and allocate training efforts, often prioritizing rapid-response capabilities over long-term efforts. Therefore, the crisis management approach can limit the effectiveness of long-term disaster risk reduction strategies.

The absence of clearly defined thresholds for identifying the start and end of a drought event further hinders proactive intervention, often resulting in delayed responses when substantial impacts have already manifested.⁴⁵ This lack of clear drought threshold levels has implications for the accountability and responsibilities of the various institutions involved in drought management.⁴⁶

As drought conditions persist and affect the population, critical infrastructure, and supply chains, the crucial role of civil protection becomes increasingly apparent. However, the absence of a clear mandate or established protocols, particularly for hazards with limited experience, can impede optimal collaboration between civil protection partners in emergency situations, resulting in delays in response efforts.⁴⁷

In many countries, civil protection and disaster management operate within a decentralized framework, involving a multi-level, multi-sector network of stakeholders across all the stages of the disaster cycle. While the decentralization of responsibility to local governments can be beneficial, it also necessitates effective vertical coordination among stakeholders at all levels of governance to ensure seamless communication, resource allocation, and decision-making during drought events.

Key Terminology

Based on the Sendai Framework Terminology on Disaster Risk Reduction⁴⁸

Prevention refers to activities and measures taken to avoid existing and new drought risks. While drought risk cannot be fully eliminated, prevention aims to reduce communities' vulnerability and exposure. This could involve implementing water conservation measures, improving land-use practices, or developing drought-resistant crops.

Mitigation involves actions taken to lessen or minimize the scale or severity of a drought, which often cannot fully be prevented. This might include engineering techniques and hazard-resistant construction as well as improved environmental and social policies and public awareness.

Preparedness indicates the increased knowledge and capacities developed by governments, professional response and recovery organizations, communities, and individuals to effectively anticipate, respond to, and recover from the impacts of likely, imminent, or current disasters. Some examples are the development of emergency management organizations to plan and coordinate response, installing early warning systems, identifying evacuation routes, and preparing emergency supplies.

Disaster Risk Reduction goal is the substantial reduction of disaster risk and losses in lives, livelihoods, health, and in economic, physical, social, cultural, and environmental assets of persons, businesses, communities, and countries.

1.4.2 Drought management capabilities in the Swiss civil protection

Recognizing the need for Switzerland's civil protection system to evolve and adapt to complex challenges such as droughts and other climate-related extremes, the FOCP, in collaboration with federal and cantonal partners, has recently conducted a comprehensive analysis of Switzerland's civil protection capabilities titled Capability Analysis for Civil Protection.⁴⁹ The analysis, referencing a drought scenario that could persist for six months, identified 39 key capabilities that are essential for effective mitigation and response. These include not only cross-cutting crisis management and civil protection capabilities but also specialized capabilities in data management, situational awareness, resource preservation, and water management logistics.

While the analysis highlights the breadth of necessary capabilities and resources required for effective drought mitigation, it also reveals significant deficiencies in the current capacity of the Swiss civil protection system. A substantial 19% of the identified capabilities in civil protection were rated as inadequate, with a disproportionate 67% of these deficiencies concentrated in the domain of leadership and management. Although 42% of the capabilities were deemed as sufficient and 39% assessed as good, the overall findings underscore the need for targeted improvements in various aspects of drought management. Specifically, the analysis points to the need for improved strategies for allocating and prioritizing water resources in times of scarcity, a clearer definition of roles and responsibilities at different levels of government, improved public awareness and education about drought risks, and increased collaboration between civil protection and the entities owning and managing resources and assets.

To effectively address the outlined gaps and improve overall drought resilience, a reassessment of existing structures, policies, and strategies within the Swiss civil protection system is necessary to better account for drought risks. This entails a paradigm shift from a reactive crisis management approach to a proactive drought risk reduction approach, both of which are depicted in **Figure 3**. This shift fully embraces the integrated risk management cycle⁵⁰ promoted by the FOCP and focuses on enhancing and improving monitoring, mitigation, and response mechanisms. It, therefore, enables decisionmakers to detect a drought early, respond promptly, and implement measures to reduce impacts before they reach a crisis stage.⁵¹

Figure 3 illustrates the civil protection actions framework in response to drought impacts, distinguishing between crisis management and risk reduction.

Graph inspired from Wilhite D. A., and Pulwarty R., S., Drought and water crises: integrating science, management, and policy, Second Edition, Boca Raton : CRC Press, 2018 p. 100



1.4.3 Swiss Civil Protection's evolution in drought management

As civil protection in Switzerland shifts from focusing on crisis management to a more proactive approach to risk reduction by implementing the gaps that have also been identified in the Capability Analysis for Civil Protection,⁵² several significant changes are anticipated. This shift aims to increase resilience and ultimately reduce the impact and costs of disasters such as droughts and other climate-related extremes. For example, studies estimate that every dollar spent on drought risk mitigation saves at least two dollars in future disaster costs.⁵³ This transition will create new opportunities as well as new demands for civil protection organizations to evolve and successfully manage complex emergencies.

One of the most immediate impacts of this shift will be a strain on resources due to more frequent and resource-intensive operations.⁵⁴ Civil protection organizations will face increased demands on personnel, material, and time.⁵⁵ Innovative solutions will be needed, such as expanding the volunteer base and leveraging spontaneous helpers organized through social platforms. Engaging communities in civil protection efforts will not only increase the number of personnel but also promote a culture of early action and resilience.⁵⁶

Moreover, prolonged and intensified deployments due to more frequent and severe droughts can have a toll on the mental health of civil protection personnel. This requires the implementation of support mechanisms and strategies to address potential psychological impacts.⁵⁷ The provision of mental health resources, stress management training, and counseling services will be essential to maintain the well-being of civil protection personnel and ensure they remain healthy and effective in their roles.

The transition to risk reduction will also ask for a reassessment of equipment needs to ensure preparedness for future requirements. This may include the procurement of specialized assets, such as mobile water treatment units and emergency power generators, or the pre-positioning of emergency supplies. However, as budget constraints may limit some cantons' ability to acquire such equipment independently, the importance of intercantonal cooperation and resource sharing through bilateral agreements and the Federal Resource Management Mechanism (*Ressourcenmanagement Bund, ResMaB*) is underscored. By pooling resources and coordinating efforts, cantons can overcome financial constraints and ensure all regions have access to essential equipment.

Maintaining the operational continuity of critical infrastructure, including energy, water, transportation, and telecommunications systems, is another key responsibility of the technical services (*Technische Betriebe*) as one partner organization within civil protection.⁵⁸ This becomes even more important during prolonged drought condi-

tions, as many of these systems are particularly vulnerable to disruptions caused by water scarcity and shortages.

Furthermore, the shift towards proactive risk reduction will require a greater emphasis on training and prevention activities. This includes developing comprehensive risk management plans that include both proactive and reactive measures. Training programs must be updated to include scenarios that focus on slow-onset disasters, such as droughts, to ensure that personnel are well-prepared to deal with these complex situations. Ongoing professional development and regular joint exercises with other civil protection entities will be crucial in building a robust and adaptable civil protection workforce.⁵⁹

At a more strategic level, preparedness efforts will include reviewing existing water distribution protocols, engaging diverse stakeholders for unified action, and preventing fragmented responses such as uncoordinated water bans. To address these challenges, a greater emphasis will be placed on improving civil protection leadership capabilities, as highlighted in the Capability Analysis for Civil Protection.⁶⁰

Within cantonal offices, civil protection is expected to strengthen its role as a subject matter expert in both strategic and operational preparedness while serving as a platform for dialogue between the various specialized agencies involved in drought management. This could involve regular coordination meetings with relevant cantonal authorities and agencies, such as water management, forestry, and public health, to proactively address and plan for potential drought-related challenges - an approach already being implemented in some cantons. By facilitating collaborative discussions and collecting recommendations to present to the cantonal executive power, cantonal civil protection authorities can foster trust and cooperation within their canton. Between cantons, platforms such as the Conference of Cantonal Authorities for the Military, Civil Protection, and Civil Defense (Konferenz der kantonalen Verantwortlichen für Militär, Bevölkerungsschutz und Zivilschutz, KVMBZ)61 and other knowledge-sharing networks should be used to elevate ambitions, drive coordinated action and enhance overall community resilience in the face of drought.

At the national level, cantonal civil protection organizations could bring valuable local and regional knowledge and practical experience in crisis management to the table, ensuring that national policies and strategies are informed by real-world challenges and lessons learned.

2 Drought preparedness in Switzerland

To gain a holistic understanding of the diverse drought management practices across Switzerland, this study employed a two-step expert engagement process. First, an online survey was distributed to 26 cantonal authorities responsible for the strategic development of civil protection within their respective cantons or involved in cantonal hazard analysis and preparedness planning. This was followed by in-depth interviews with representatives from civil protection and defense offices in six selected cantons – Basel-Stadt, Thurgau, Zurich, Bern, Fribourg, and Ticino – as well as national-level experts from Meteo-Swiss, FOEN, LAINAT, and WSL. These interviews provided deeper insights into past drought events, current challenges, the effectiveness of existing measures, and ongoing national drought mitigation efforts.

The findings from this engagement offer a snapshot of current drought practices in Switzerland, revealing strengths and identifying areas for improvement. They have been instrumental in shaping the Drought Mitigation Toolkit presented and validate the 10 Action Points in this report. Moreover, the expert input offers a forward-looking perspective on planned projects and initiatives aimed at enhancing drought resilience at both the cantonal and federal levels.

2.1 Survey results

This section provides an analysis of drought preparedness across Swiss cantons based on quantitative data gathered from the survey's structured questions. Given that civil protection in Switzerland is primarily organized at the cantonal level, the survey aimed to capture baseline data on existing strategies and procedures, offering a snapshot of the cantonal approaches to drought management. Additionally, it sought to assess overall preparedness through the use of a hypothetical drought scenario.

The survey had participation from 25 out of 26 cantonal civil protection authorities from the cantonal Office of Military and Civil Protection/Civil Defense. This near-complete dataset allows for a comprehensive analysis of drought preparedness across Switzerland's diverse geographic, linguistic, and regional landscape. A detailed overview of the survey results can be found at the end of this Section.

A noteworthy finding from the survey is the pervasive acknowledgment of **drought risk** among cantonal civil protection offices, with 24 cantons rating it as medium to high for their respective canton (Question 1). This highlights the increasing recognition of drought as a significant threat to Switzerland, irrespective of regional variations. Furthermore, 16 cantons have incorporated drought into their hazard analyses, typically in conjunction with heatwaves or wildfires as related hazards (Question 3). This number is likely to increase as some cantons are in the process of updating their cantonal hazard analysis. In fact, 9 cantons' hazard analyses are dated prior to 2018 (Question 2). However, the prevalent association of drought with heatwaves suggests that the drought impacts are primarily recognized during the summer months, coinciding with higher temperatures. This perspective may inadvertently overlook the critical role of winter conditions such as snow scarcity or low snow precipitation, which are crucial for water supplies, agriculture, hydropower production, and ecosystems throughout the year. Further investigation is required to ascertain whether this association restricts drought management to the summer season only.

A total of 19 cantons indicated that they experienced drought events in the past (Question 5). These experiences have prompted a range of responses, with 8 cantons conducting informal evaluations of these events and 5 undertaking comprehensive evaluations that yielded specific and actionable recommendations (Question 6).

The implementation of post-drought adaptation measures has resulted in the introduction of specific changes at the cantonal level, as reported by 18 respondents (Question 7). These include the establishment of drought-focused working groups or task forces in 15 cantons and enhancements to water storage and distribution infrastructures in 10 cantons. In the absence of the planned national drought early warning system at the time of the survey, 8 cantons have independently refined their monitoring and early warning systems in response to previous droughts.

As of May 2024, only 7 cantons had developed specific drought management plans, with 1 of these cantons in the process of doing so (Question 8). This highlights a gap in the overall national capacity to manage droughts. Existing plans frequently adopt a multi-hazard approach, addressing not only drought but also the interconnected risks of heatwaves, wildfires, the spread of invasive species, or air pollution (Question 11). Such plans typically incorporate a range of response measures, including the pre-established emergency protocols, the formation of dedicated drought task forces, clearly defined roles and responsibilities for civil protection partners, and ongoing drought monitoring activities. However, the absence of key elements such as proactive mitigation strategies, postdrought evaluation processes, updated analyses of historical drought impacts, or a clear identification of the sectors most vulnerable to drought represents a notable gap in the existing drought management plans. Furthermore, 4 of the 7 drought management plans lack clear criteria for declaring drought emergencies, which is a crucial element in ensuring timely and effective responses (Question 10).

So far, 15 cantons do not regularly publish a **drought bulletin** (Question 12). This observation underscores the potential benefits of the planned national early warning system and its associated national weekly drought bulletins.

Furthermore, 16 cantons have appointed a **drought expert or task force**, often led by specialists from the cantonal environmental or energy offices. This demonstrates an acknowledgment of the necessity for specialized knowledge in addressing drought-related challenges (Question 13).

The implementation of scenario-based **exercises or training** is an indicator of operational preparedness. However, 21 cantons lack this for drought scenarios (Question 15). **Cross-border collaboration** with neighboring cantons or countries, while under consideration in 16 cantons, remains limited (Question 14). This response highlights a significant area of concern, as droughts do not respect political boundaries and frequently impact extensive geographical areas.

Concluding the first part of the survey, 14 cantons have not yet initiated research and development programs to improve drought resilience within their canton. This finding indicates a gap in proactive measures and long-term planning or resource constraints for drought management (Question 16).

In the second phase of the survey, respondents assessed their canton's preparedness to address a drought scenario. With 17 cantons, a majority indicated "moderate preparedness", and 7 expressed an even lower level of readiness for the presented scenario (Question 17). This self-assessment raises concerns about the overall capacity of Switzerland to effectively respond to drought events. Despite this, 23 respondents attested that their cantonal civil protection possesses a sufficient legal basis to fulfill their mandate of protecting the population and its vital resources in the event of a drought emergency (Question 18). Additionally, 16 cantons indicated that they have implemented adequate strategies and plans to manage water shortages (Question 20). However, a recurring concern was the absence of an overarching framework that integrates all relevant actors, particularly municipalities responsible for local water management. Furthermore, challenges were identified with regard to public engagement in water conservation efforts and a lack of clear responsibilities and communication strategies at all decision-making levels (Question 21).

As indicated by the respondents, the three sectors that are expected to be most affected by drought are agriculture, drinking water supply, and energy production (Question 22). In order to raise awareness and promote water conservation, cantons employed a variety of strategies, including extensive media coverage, informational websites, social media campaigns, and targeted water conservation initiatives (Question 23). The respondents identified several pivotal challenges associated with the management of drought scenarios in their respective cantons, with the majority of these challenges pertaining to the domain of **water management**. These challenges include the adequacy of infrastructure for water storage and distribution, the balancing of water usage for irrigation, industrial, and public supply needs, and the enforcement of water use restrictions and conservation measures. Securing financial resources for the implementation of mitigation measures also poses a notable hurdle (Question 24).

The allocation of **financial resources** to bolster drought response capabilities is primarily based on the cantonal budget, followed by federal assistance (Question 25).

Only three cantons have implemented **staff and personnel training** on drought preparedness and response, and one is in the process of doing so (Questions 26 and 27).

The survey results also point to promising developments. Despite the currently limited **cross-border collaboration** regarding drought management, 10 respondents expressed interest in engaging at both intercantonal and international levels, while 8 respondents indicated interest in inter-cantonal engagement only (Question 28). Furthermore, the considerable interest in participating in **a national knowledge exchange program** on drought management and resilience highlights a desire for shared learning and collaborative solutions (Question 29).

Section 1

1. How does your office estimate the risk of drought in your canton over the next ten years? (n=25)



2. When was your canton's hazard analysis last updated? (n=24)



3. How does your canton's hazard analysis address drought? (n=25)

Drought is included in the hazard analysis, but it is combined with a related hazard (e.g., drought and heatwave).

Drought is recognized as a specific hazard and is included in the hazard analysis.

While drought specifically is not directly included in the hazard analysis, heatwave is addressed.

The hazard analysis does not specifically address drought or any related hazards.

1

Other

If Other, please specify:

- Not following the national classification of the National Risk Analysis of Disasters and Emergencies in Switzerland. 2

4. What are the reasons for not specifically addressing drought in the canton's hazard analysis? (n=3)

- Drought will be included in the future update of the hazard analysis. 2
- In the process of development of the hazard analysis, drought was categorised as a non-priority hazard for the canton. 1

5. Has your canton experienced drought events in the past 20 years? (n=25)

Yes	19
No 3	
Unsure 3	

6. Following the drought event that impacted your canton, was there a documentation and evaluation process to assess the effectiveness of drought response actions? (n=19)

An informal evaluation process was conducted without structured documentation.

A comprehensive evaluation process was conducted, resulting in actionable recommendations.

No formal evaluation process was conducted.

Other

7. Following the drought event that impacted your canton did your canton incorporate any of the following changes? (n=18)

Creation of a drought working group/task force

Enhancement of infrastructure for water storage and distribution

Refinement of the monitoring and early warning system

Revision and update of the hazard analysis

Inclusion of scientific advisory into the drought management

Elaboration of cross-sectoral strategies to address water scarcity

Establishment of a drought management plan

No changes were implemented

2

Increase of research and development to improve drought resilience

1

Introduction of educational programs and awareness-rising campaigns

1

Other

••• 3

If Other, please specify:

- Trink- und Brauchwasserplanung und Dokumentationen. 2
- Institutionalisierung der Lageanalyse und Massnahmenergreifung. 1

8. Does your canton have a drought management plan? (n=24)



9. Since when does your canton have a drought management plan? (n=7)



10. Which key elements does the drought management plan include / will include? (n=7)



11. Does the drought management plan also address the following drought-related hazards? (n=7)

Heatwave

Wildfires

Spread of invasive species (e.g. Bark Beetle)

Power failure/Power shortage

Low air quality

2

Restriction of navigation on waterways

Shortage of agricultural goods

1

12. Does the Canton publish a regular drought bulletin? (n=24)



Planned

Yes, and it is publicly available

Yes, but it is not publicly available

If Planned, please specify a timeframe: - 2025 2 13. Does your canton have a drought expert/task force? (n=24)





If Planned, please specify a timeframe:

- 2025 2
- 2024–2025 1

14. Is your canton currently involved in addressing drought through cross-border collaboration with neighbouring cantons or countries? (n=23)

No involvement in any cross-border activities

Yes, involved in informal engagement

Yes, through established formal channels

No, but cross-border collaboration is under planning

1

15. Has your canton engaged in, or does it plan to engage in, drought scenario-based exercises and/or training? (n=25)





Planned

0

16. Has your office initiated research or development aimed at improving drought resilience in your canton? (n=24)



Section 2

Drought Scenario Exercise

It is early March, and the seasonal outlook for the coming summer indicates that drought conditions could severely impact Switzerland. The country is already experiencing long-lasting, above-average temperatures combined with significantly low precipitation levels. These are affecting soil moisture and vegetation growth, crucial for Switzerland's ecosystems and agriculture. In particular, countries bordering Switzerland have already experienced significant impacts on their water resources due to prolonged drought conditions. Your Canton is at risk of facing significant water scarcity, potentially affecting local communities, economies, and ecosystems.

17. How would you rate your canton's current level of preparedness for the drought scenario outlined above? (n=24)

Very prepared 0 Moderately prepared Somewhat prepared **00000**5

Minimally prepared 2

18. To what extent do you agree or disagree with the following statement: "The cantonal civil protection has a sufficient legal basis to fulfil its mandate of protecting the population and its vital resources in the event of a drought emergency." (n=25)

Agree

Disagree 2

19. If you disagree, what do you believe is missing in the current legislation? (n=2)

- The process of implementing the revision of the Federal Law on Civil Protection and Civil Defense (520.1 Bundesgesetz über den Bevölkerungsschutz und den Zivilschutz) at the cantonal level is either underway or it should be improved. 2

20. To what extent do you agree or disagree with the following statement: "Our canton has sufficient strategies, plans and measures in place to manage potential water shortages." (n=24)





21. If you disagree, what do you believe is missing in your canton in order to manage potential water shortages? (n=7)

- A clear division of responsabilities. 1
- An emergency plan in case of water shortage. 1
- A strategy or an overall framework that includes and comprehensively addresses stakeholders, communes (which in Switzerland are responsible for water supply), regional, cantonal, or extra-cantonal levels. 1
- A clear communication strategy aimed at saving water resources upstream. 1
- The readiness of the population to implement measures to reduce water wastage. 1
- An emergency plan. 1

22. Given the scenario, which sectors in your canton do you anticipate being most affected by drought conditions? (maximum THREE options can be selected) (n=24)

Agriculture

Public water supply

Energy production

Fishery

Healthcare and public safety

Timber

Other, please specify

Recreation and tourism

Navigation and shipping

23. Which public awareness strategies has your office implemented to improve drought awareness and water conservation efforts in your canton? (select all that apply) (n=18)

Extensive coverage in local newspapers

Creation of an informational website or online portal

Active engagement on social media via official cantonal channels

Water conservation campaigns (e.g., Water Awareness Week)

Targeted workshops for specific groups or sectors

Educational programs in schools

Community events aimed at raising environmental awareness

0

Collaboration with local businesses

0

Installation of public displays or exhibitions ^o

Other



24. Please assess the extent to which the following aspects could present challenges for your canton in managing the drought scenario. (1 = Minimal challenge, 5 = Very high challenge) (n=25)

25. In terms of response capacity to the drought scenario, where would the primary source of support come from? (more than one option can be selected) (n=23)

Cantonal Budget

21

Federal assistance

Public private partnerships

Municipality budget

26. Has your office conducted staff training on drought preparedness and response? (n=24)



Yes

Planned

27. If your office has conducted or is planning to conduct staff training, which of the following activities have been implemented, or will be implemented? (Select ALL that apply) (n=4)

Participation in multi-agency exercises

Emergency drills

External workshops and courses

1

Field visits to areas previously affected by drought events

1

Intercantonal exchange programs

1

Community engagement initiatives

0

Other

0

28. Would your canton consider engaging in inter-cantonal or international cooperation efforts to address drought management? (n=24)

Yes, both inter-cantonal and international

Yes, inter-cantonal only

No

Yes, international only

0

29. Would your canton be interested in participating in a national civil protection knowledge exchange program on drought management and resilience building? (n=24)



2.2 Expert interviews results

This section synthesizes the insights gained from the expert interviews on drought mitigation efforts across Switzerland. Although the individual interviews are not published, their findings encompass key theme key themes, shared challenges, and opportunities in drought management at different administrative levels.

 Institutional Framework and Coordination: A complex network of institutions is involved in drought management in Switzerland, encompassing federal agencies such as the FOCP, the FOEN, Meteo-Swiss, and Federal Office of Topography (Swisstopo), alongside cantonal governments and municipalities. Coordination among these entities is essential but often challenging due to limited exchanges on drought-related issues, which can lead to siloed actions. The establishment of drought task forces and working groups at both federal and cantonal levels reflects a growing recognition of drought as a significant issue that requires dedicated attention and resources.

Past drought events have acted as catalysts for creating these working groups and integrating drought considerations into cantonal hazard analyses. At the federal level, there are high expectations surrounding the restructuring of crisis management frameworks, including the establishment of a dedicated specialist team focused on drought (*Fachstab Trockenheit*). However, the specific role of the FOCP in this new structure is still being clarified.

At the cantonal level, drought working groups typically operate during summer months, highlighting their reactive nature in hazard management. These groups rarely function year-round, which limits proactive planning. While cantonal civil protection authorities are integral to these groups, there remains an opportunity for them to effectively convene and coordinate these groups.

- Role of Civil Protection: Civil protection agencies are increasingly recognized for their crucial role in drought management at both federal and cantonal levels, particularly in coordinating responses and communicating alerts and measures to the public. However, the specific responsibilities and extent of their involvement in drought preparedness and response can vary significantly between cantons. This inconsistency reflects broader differences in approaches to drought management, levels of preparedness, and perceptions of drought risk across Switzerland. Some cantons have developed comprehensive drought management plans, while others are still in the early stages of addressing this issue. Platforms such as the Conference of Cantonal Authorities for the Military, Civil Protection, and Civil Defense or joint exercises in civil protection provide opportunities to exchange knowledge and advancements in drought mitigation.
- Challenges and Needs: Managing past drought events in Switzerland has highlighted several key challenges and areas for improvement. One of the foremost issues is the need for stronger coordination between all levels of government – federal, cantonal, and municipal. Equally pressing is the need to raise public awareness about the complexity of drought, its long-term impacts, and the importance of water conservation. A lack of public understanding can weaken drought management efforts and may also lead to water conflicts.

Moreover, there is a widespread misconception that drought management can be handled solely within cantonal borders, overlooking the interconnected nature of water resources. Decisions in one canton can have far-reaching consequences, affecting water availability and management in others.

Another critical challenge is the development of more precise and localized drought indicators, a gap that the upcoming national drought monitoring and early warning system aims to address. There is also a clear need for more research into drought patterns, impacts, and management strategies tailored to Switzerland's unique geography and climate.

Enhanced networking and collaboration among experts, decision-makers, and stakeholders across sectors is considered to be crucial for developing integrated drought management strategies. Given Switzerland's central European location, cross-border cooperation is increasingly important. However, this cooperation is often hindered by differing administrative structures, conflicting priorities, and varying water management policies in neighboring countries. **Early Warning System Development:** Switzerland's drought monitoring and early warning system aims to replicate the success of similar systems for other natural hazards like floods. The initiative presents significant opportunities, including enhancing coordination among federal, cantonal, and municipal bodies, as well as providing real-time data to support decision-making across sectors such as agriculture and water management. The system also plans to incorporate advanced monitoring tools like the soil moisture network and satellite data, offering precise regional forecasts. However, the system's success will depend on its ability to provide timely, accurate warnings while respecting Switzerland's federal structure and diverse regional needs.

• Long-term Resilience: Many interviewees emphasized the importance of developing long-term strategies for drought resilience, considering factors such as climate change, population growth, and changing water use patterns. This includes not only reactive measures but also proactive steps to adapt water management practices and infrastructure.

3 Drought mitigation toolkit: 10 Action Points



6. Appoint a drought task force

The Drought Mitigation Toolkit offers a framework for developing regional drought management policies and preparedness plans. The toolkit has been designed to reflect the diverse realities of cantonal civil protection organizations in Switzerland and to be sufficiently flexible to meet different local needs, with the aim of enhancing drought resilience. Although the toolkit has been designed with the specific context of Switzerland in mind, its underlying methodology is applicable and adaptable to various levels of government and authorities worldwide.

The toolkit comprises 10 Action Points, identified as core to its functionality following the analysis of the survey results and expert interviews in Section 2. These Action Points are designed to address gaps in preparedness identified through the analysis and include:

- Action Point 1 Establish and incorporate a national drought early warning system
- Action Point 2 Assess drought risk
- Action Point 3 Include drought in the cantonal hazard analysis
- Action Point 4 Translate drought risk to local action
- Action Point 5 Develop a drought response plan
- Action Point 6 Appoint a drought task force
- Action Point 7 Strengthen drought preparedness through exercises
- Action Point 8 Participate in knowledge exchange
- Action Point 9 Conduct community education and outreach programs
- Action Point 10 Embrace collaboration

The Action Points offer general guidance for the development of drought management strategies, focusing strongly on early action and mitigation. The toolkit is consistent with the integrated risk management approach advocated for Swiss civil protection.⁶² As potential measures against drought are often of a long-term, preventive character, the 10 Action Points provide resources for the development of long-term mitigation strategies, as opposed to short-term response actions. Each Action Point includes illustrative examples and references to drought management practices from Swiss cantons, thereby showcasing the diversity of existing initiatives. The objective of the toolkit is to facilitate the exchange of resources and to highlight common challenges in order to foster inter-cantonal communication and, ultimately, inspire action.

A number of resources are available at the international, national, and regional levels and offer guidance on how to shape drought strategies. The most prominent one is the "10 Steps National Drought Management Policy," which was developed by the Integrated Drought Management Programme, a platform developed by the World Meteorological Organization in collaboration with the Global Water Partnership.⁶³ Another valuable resource is the United Nations Convention to Combat Desertification's Drought Toolbox, which provides drought stakeholders with easy access to tools, case studies, and other resources to facilitate the design of national drought policy plans and support the achievement of the United Nations Sustainable Development Goal 15 (Protect, restore and promote sustainable use of terrestrial ecosystems, sustainably manage forests, combat desertification, and halt and reverse land degradation and halt biodiversity loss).⁶⁴

These resources have been vital in guiding the development of the Action Points in this toolkit that form a roadmap to drought resilience in Switzerland. By considering their indications, the Action Points ensure that the Swiss roadmap is aligned with the national strategies implemented by other countries.

The Action Points in the toolkit are not ranked in order of importance, as each contributes a unique element to strengthening drought resilience. However, recognizing that implementation requires resources, civil protection organizations are encouraged to prioritize Action Points based on their specific needs and circumstances. Moreover, certain Action Points will require efforts in another Action Point (e.g., the translation of drought risk to local action assumes that drought risk has been analyzed in one form or the other), which is indicated in the corresponding Action Points.

Lastly, the toolkit is conceived as a living document and dynamic process, anticipating that new information, inputs, and case studies will prompt an iterative approach to ensure that the toolkit remains relevant and reflective of the evolving institutional, social, and environmental landscapes of Switzerland.



Action Point 1 – Establish and incorporate a national drought early warning system

Drought monitoring and early warning systems (EWS) constitute the cornerstone of proactive drought management,⁶⁵ as they shift the focus from reactive crisis management to anticipatory action. These systems provide timely and accurate information regarding the development of drought conditions to decision-makers, stakeholders, and the public, thereby enabling them to take proactive measures to reduce the impact considerably. The goal of Action Point 1 is to illustrate how the establishment of a national-scale drought EWS amplifies these benefits by ensuring standardized monitoring, enabling a coordinated response across cantons and sectors, optimizing resource allocation, and fostering evidence-based drought policy development. This approach aligns with the "Early Warnings for All" Action Plan,66 which was developed by the United Nations. The plan highlights the critical role of early warning systems in building resilience to drought and reducing its adverse socio-economic and environmental consequences.

With regard to hydrological extremes in Switzerland, due to past experiences, there has been a greater focus on managing flood events. In particular, the 2005 floods served as a driving force for substantial improvements in the national warning and emergency intervention processes.⁶⁷ However, the attention towards establishing a drought warning system has been slower in making its way onto the political agenda. As indicated in a 2018 *OWARNA* (Optimierung von Warnung und Alarmierung) report from the *Steering Committee Intervention against natural Hazards (Lenkungsausschuss Intervention Naturgefahren-LAINAT)*, the absence of a national drought monitoring system has resulted in the emergence of several challenges.⁶⁸

First, there is a lack of consensus regarding the definition of drought across cantons and sectors. Each actor defines drought based on its primary concerns, resulting in a fragmented understanding of this phenomenon. The FOCP has addressed this challenge in the National Risk Analysis by providing a baseline definition for drought and a scenario that cantonal civil protection authorities can utilize as a foundation for their own hazard analyses.⁶⁹

Second, most of the existing warnings tend to prioritize immediate threats to life and property,⁷⁰ while overlooking the prolonged nature of droughts. This ne-

Drought monitoring and early warning systems (EWS) facilitate proactive management by providing users with timely and accurate information, thereby enabling a shift from reactive to anticipatory action. A national-scale EWS for drought will standardize monitoring, support the coordination of responses across cantons and sectors, and strengthen evidence-based policy development in alignment with the UN's "Early Warnings for All" Action Plan.

cessitates a novel type of forecasting and warning format that integrates various specialized disciplines.

Third, inconsistent response efforts across sectors and cantons remain a challenge, as drought warnings and related measures are often restricted to the administrative boundaries of individual cantons or municipalities.

In response to the recommendations set forth in the *OWARNA* report, which emphasized the necessity for a robust drought warning system, the Federal Council, on May 18, 2022, assigned to FOEN, MeteoSwiss, and Swisstopo the task of developing a unified national drought monitoring and early warning system by 2025.⁷¹ This initiative aims to integrate a range of resources to address the complex and multifaceted nature of drought, extending beyond the domain of agricultural drought. Furthermore, the initiative strives to harmonize and standardize drought-related information across the entire Federation, thereby ensuring a coherent and consistent understanding of what constitutes a drought.

The drought warning system will serve a broad user base, including cantonal and local authorities, as well as sectors that are impacted by drought, such as agriculture, hydropower, drinking water supply, shipping, nature conservation, and insurance. This system will provide an enhanced understanding of drought conditions in 36 Swiss regions by using environmental indicators and a combined drought index. It will help users across different sectors and administrative levels to implement preventive measures better, thus preventing damage and minimizing shortages of basic services.⁷² The system will incorporate several key components, as depicted in **Figure 4**, forming a comprehensive drought warning chain:

- Drought Definition: Establishing a clear, nationally consistent definition of drought, warning thresholds, and integration into the existing danger levels.
- Monitoring: Utilizing existing meteorological and hydrological measurement networks and incorporating soil moisture stations.
- Analysis and Forecasting: Developing an operational analysis and forecasting system based on a combined drought index calculated on a daily basis for each of the 36 Swiss regions, with predictions extending over a four-week period.

 Communication and Outreach: Disseminating a weekly national drought bulletin, similar to the Natural Hazards Bulletin, which provides comprehensive information on the current drought situation and forecasts in Switzerland.

This approach of standardization will facilitate the dialogue and collaboration between cantonal authorities and the Federal Government by establishing a consistent baseline for discussions and decision-making processes. The introduction of a national drought EWS has been met with general acceptance by cantonal civil protection authorities, as evidenced by the findings of the conducted expert interviews in Section 2.2. The system can assist cantonal civil protection authorities in communicating effectively and consistently, utilizing uniform reference points, by providing an initial classification of drought conditions based on a standardized framework. This shared understanding will ensure that all stakeholders are operating with the same information, irrespective of the canton or sector. In conclusion, this should result in more coordinated and effective drought management strategies, thereby contributing to the overall enhancement of civil protection capabilities in Switzerland, as targeted in the Capability Analysis for Civil Protection.73

The development of a national drought EWS can also strengthen Switzerland's institutional capacity to deal with slow-onset or long-lasting crises. To date, the primary focus of national specialist staff, such as the Natural Hazards specialist staff *(Fachstab Naturgefahren),* has been the coordination of short-term warnings for acute events. Their secondary role as advisory bodies, providing necessary knowledge and expertise to federal staff during prolonged crises, has remained largely untapped. The national drought EWS offers a potential avenue for integrating these two functions, namely short-term coordination and long-term advisory capacities.⁷⁴

The efficacy and applicability of the national drought EWS will be assessed through an iterative process of improvements, forecast verification, and user feedback. The long-term objective is to extend the forecast horizon and enhance the accuracy and reliability of drought forecasts.

However, the development and implementation of this system are confronted with several challenges, as outlined in Section 2.2. First, ensuring the scientific precision of drought forecasts requires rigorous validation and ongoing refinement of prediction models. Second, effective coordination between the cantonal and federal levels is essential for a unified response. Striking the right balance between national guidance and respecting cantonal autonomy, however, can be difficult. Interviews with cantonal civil protection authorities have revealed that while the system's monitoring, prognosis, and forecasting capabilities are welcomed, the responsibility Figure 4 shows the components included in the upcoming Swiss national drought early detection and warning system



for implementing drought measures ultimately lies with the cantons. Therefore, drought warnings and related recommendations must be carefully crafted to reach the affected individuals and organizations and to avoid conflict with existing cantonal measures, which can create issues in the messaging process and successive communication to the public.⁷⁵

Lastly, expanding the drought analysis to include neighboring regions in bordering countries would provide valuable insights into cross-border dynamics. However, this requires a political mandate to facilitate data harmonization and cross-border collaboration, as outlined in Action Point 10 "Embrace collaboration".

In conclusion, the establishment of a national drought monitoring and EWS signifies a substantial advancement for Switzerland in its endeavors to address the increasing risk of drought. This EWS will serve as a foundation for several Action Points, which are successively presented in this report. These include assessing drought risk, developing drought response plans, raising community awareness about droughts, and ultimately building a more drought-resilient Switzerland.



Action Point 2 – Assess drought risk

Drought is a hazard that encompasses both environmental and societal dimensions. The environmental aspect of drought relates to its probability and severity in a given region. However, it is the social dimension, specifically the vulnerability of societies and their infrastructure, that transforms a hazard into a risk. If this risk is manifested, it may potentially become a disaster.⁷⁶ Action Point 2 addresses this social dimension in the context of conducting a drought risk assessment, using the Canton of Basel-Stadt as a reference for the discussion. This assessment should identify the ways in which drought hazards manifest and impact both natural and social systems, thereby enabling civil protection organizations to more accurately determine intervention thresholds and develop effective mitigation strategies.

A comprehensive assessment of drought risk is a complex undertaking, particularly given the cascading and compounding effects that even moderate droughts can have on interconnected socio-ecological systems.⁷⁷ Nevertheless, a risk assessment is indispensable for determining when a drought hazard requires intervention from civil protection organizations, whether preventive or in response. The insights and recommendations presented here aim to inform these discussions and support other Swiss cantons initiating or refining their drought risk assessment processes.

Historically, Basel-Stadt has not perceived the exposure to drought as a significant threat, given that the canton's landscape is primarily urban and agricultural activity is relatively limited. The ever-flowing Rhine River⁷⁸ further reinforced this perception, directing attention toward flood management rather than drought preparedness. However, the recent increases in extreme weather events have heightened concerns about the canton's susceptibility to climate-related extremes. Although the Canton of Basel-Stadt is not directly affected by agricultural drought, its position as a key hub on the Rhine River renders it vulnerable to the logistical challenges that emerge during periods of low water levels. For instance, the decline in container traffic at Rhine port terminals in 2023 was partially attributable to unfavorable water levels,79 with the economic consequences potentially spreading to Switzerland as a whole. This is because freight transport by water in Switzerland is essentially limited to shipping on the Rhine, with around 10% of all Swiss imports and exports in 2023 being hanDrought is a complex phenomenon with both environmental and social dimensions, where the social vulnerability can often transform the hazard into a disaster. The Canton of Basel-Stadt shows that even regions with minimal exposure could experience significant drought impacts if vulnerabilities are not assessed.

dled in the Rhine ports of Basel-Stadt and Basel-Landschaft.⁸⁰ The Rhine is arguably the most closely monitored and regulated river in Switzerland. It has been meticulously mapped to facilitate a rapid crisis response (*Einsatzplanung Rhein* project), and alarm systems have been established to alert the relevant authorities in the event of high water levels. Yet, a comparable framework for low water levels is absent. In contrast to the suspension of shipping during floods, there is no defined threshold for low water levels that could result in the cessation of maritime transportation despite the potential impact on national logistics.

Moreover, while drinking water shortages caused by drought are not a threat to Basel-Stadt, high water temperatures can pose an issue.⁸¹ Although Rhine water temperatures above 25°C are currently rare and limited to the months of July and August,⁸² they could become more frequent in the future due to a changing climate. Water temperatures exceeding 25°C in the Rhine could trigger a significant spike in overall water demand, as withdrawals for industrial cooling are prohibited above this threshold. This restriction may force industrial companies to reduce power plant output^{83, 84} or to rely on drinking water for cooling purposes,⁸⁵ potentially resulting in water conflicts. Combined low water volumes and high temperatures could also severely impact fish populations and other aquatic life, particularly in the minor Birs and Wiese rivers of Basel-Stadt.86

The case of Basel-Stadt, an urban canton with historically low perceived drought risk, illustrates that even regions with seemingly minimal exposure to drought can experience significant sector-specific impacts. This vulnerability is shaped by societal factors such as community preparedness, existing infrastructure, and resource allocation within a region.⁸⁷ Thus, a comprehensive drought risk assessment must include the vulnerability component as shown in Figure 5 to identify the underlying social, economic, and environmental drivers that determine who and what is at risk and why.⁸⁸ Moreover, vulnerability is dynamic, evolving with societal changes such as population shifts, technological advancements, and policy adjustments. For this reason, regular risk reassessments are critical to capture these dynamic changes and ensure that risk mitigation strategies remain effective.

Figure 5 illustrates the three key components that contribute to risk assessment: hazard, exposure, and vulnerability.

Source of the definitions: Disaster Risk Reduction Terminology, UNDRR 2017

Exposure: Presence of people, infrastructure, housing, production capacities and other tangible human assets located in hazard-prone areas.

Hazard: A process, phenomenon or human activity that may cause loss of life, injury or other health impacts, property damage, social and economic disruption or environmental degradation.

Risk Assessment

Vulnerability:

The conditions determined by physical, social, economic and environmental factors or processes which increase the susceptibility of an individual, a community, assets or systems to the impacts of hazards.

Recognizing the importance of vulnerability factors in risk management, the Cantonal Crisis Organization of Basel-Stadt (*Kantonale Krisenorganisation*) is currently developing a resilience concept, as mandated by the Cantonal Council (*Regierungsrat*).⁸⁹ This concept will define concrete actions to strengthen the canton's resilience to different crisis scenarios, including drought. It will also identify areas where resource allocation, particularly towards initiatives like business continuity management and organizational development, can substantially improve resilience.

Furthermore, due to the wide-ranging direct and indirect (and often cascading) impacts of droughts, risk assessments must be tailored to specific sectors and user needs.⁹⁰ For cantons initiating these assessments, priority sectors include agriculture, energy and industry, potable and domestic water supply, navigation, ecosystems, tourism, forestry, aquaculture and fisheries, and the financial sector (investors, insurers, asset owners).⁹¹ The population's reliance on each sector – for income, food supply, water, and electricity – along with the economic value of each sector are critical factors in determining the canton's vulnerability to drought impacts. The National Risk Analysis assesses these sectors through the development of 12 damage indicators, including individuals, environment, economy, and society.⁹²

The experience of Basel-Stadt also highlights that drought impacts can extend beyond regional borders, potentially becoming a national issue. Therefore, the far-reaching effects of drought often transcend sectoral and governmental boundaries, emphasizing the need for cooperation and coordination between different offices and stakeholders.

In conclusion, the Basel-Stadt case study serves as a reminder of the importance of a comprehensive and thorough drought risk assessment that considers both environmental and societal factors. Despite a historically low perceived risk of drought, significant impacts can arise if vulnerabilities are not thoroughly assessed. This assessment of the components of drought risk sets the stage for Action Point 3, which emphasizes the importance of incorporating drought in the cantonal hazard analysis.



Action Point 3 – Include drought in the cantonal hazard analysis

Once a canton has identified the risk components of drought – specifically, the hazard, the exposure, and the vulnerability of people, assets, and sectors – the next critical step is to integrate it into the cantonal hazard analysis. This integration ensures the allocation of specific resources and plans for effective hazard management. Action Point 3 provides an overview of the process of incorporating drought in the cantonal hazard analysis, using the Canton of Zurich as a case study to illustrate the actions involved and the importance of a thorough and inclusive process.

The cantonal hazard analysis is a comprehensive, systematically derived, and risk-based assessment of relevant hazards within a canton.⁹³ As the cantons are primarily responsible for managing disasters and emergencies, this analysis is essential for hazard mitigation planning and provides a basis for cantonal preparedness and civil protection planning. Although cantons are not legally required to conduct a hazard analysis or to follow the cantonal hazard analysis guidelines *(Leitfaden Kataplan)* developed by the FOCP, they are strongly encouraged to do so.⁹⁴ These guidelines provide a common understanding and a uniform approach to identifying hazards and associated risks.

However, the process of developing a cantonal hazard analysis reflects the political, societal, and economic landscape of each canton, while not all cantons use the FOCP's cantonal hazard analysis guidelines. A canton's hazard analysis is supposed to be updated on a reasonable periodical basis, although some cantons have not updated their analyses since before 2018, as indicated by the survey results in Section 2.1. Conducting a cantonal hazard analysis requires significant resources, including financial investment, time, and the involvement of all relevant parties through a working group to assess each hazard. Specifically, the creation of a working group is essential to legitimize the analysis to a broader audience and to integrate operational and strategic levels of cantonal planning.

In the Canton of Zurich, the hazard identification process is legally anchored in the Cantonal Civil Protection Act *(kantonales Bevölkerungsschutzgesetz)*. This law defines a hazard as relevant for civil protection intervention when an emergency or disaster overwhelms normal coping mechanisms, endangering people, assets, and In the Canton of Zurich, while drought had been discussed in previous hazard analyses, which are anchored in the Cantonal Civil Protection Act, it was formally integrated in 2021, after a thorough assessment by the hazard analysis working group. This integration led to the development of robust drought management processes that include prevention, coordination, and communication.

essential services (adapted from Art. 2 of the Cantonal Civil Protection Act of the Canton of Zurich).⁹⁵

While drought was first discussed as a relevant hazard for the canton in 2015, it was formally included in the 2021 iteration of the cantonal hazard analysis. The combined risk of drought and heatwaves was classified as the second highest risk in the canton due to their high probability and impact on the environment and the population. The reference scenario is a prolonged dry period lasting six months with an additional heatwave exceeding 35°C, potentially resulting in deaths, numerous people requiring care, and significant environmental and economic damage.⁹⁶

Three key factors drove the inclusion of drought in the 2021 cantonal hazard analysis, as was explained during the expert interview process with the representative of the civil protection risk management responsible for the cantonal Office of Risk Management (Risikomanagement Bevölkerungsschutz) of the Canton of Zurich. First, the nationwide pilot project, "Adaptation to climate change",97 prompted civil protection organizations to assess how these measures would affect their own operations. This assessment was critical to understanding the broader impact of climate change on the structure and responsibilities of civil protection, ensuring that all partner organizations were prepared for the evolving risk landscape.98 Second, health concerns related to drought and heatwaves were raised by the cantonal health department, stressing the increasing need to manage people's health conditions in extreme scenarios. Third, previous experiences with water rationing in the region have sparked significant concerns about the availability of water for essential purposes, including drinking water, cooling systems, and agricultural activities such as dairy farming and vegetable cultivation.

Notably, the work for the 2021 cantonal hazard analysis took place during the COVID-19 pandemic. Despite the challenges posed by the pandemic, workshops were held digitally, bringing together 30 individuals from the cantonal administration, municipalities, and the private sector to examine various threats and build resilience in the Canton of Zurich. Many of these individuals were also deeply involved in the management of the pandemic, which strengthened the working group's ability to analyze complex crises. This dual involvement ensured that valuable lessons learned from dealing with COVID-19 were seamlessly integrated into the hazard analysis.

The integration of drought into the cantonal hazard analysis has not only highlighted the need for clear processes in managing this hazard but has also led to the development and implementation of these very processes. This encompasses preventive actions, effective coordination among specialized agencies and cantonal offices, and streamlined communication with the public. These efforts have resulted in the institutionalization of drought management within the Canton of Zurich.

A collaborative working group comprising a diverse range of experts, from environmental scientists to emergency responders, has played a central role in this institutionalization. The group has outlined specific measures to be implemented in the event of a drought, ensuring clear roles and responsibilities for all stakeholders. It was led by the civil protection risk management team within the police organization of the Canton of Zurich, serving as a crucial platform in the canton and facilitating the connection between specialists involved at both the operational and strategic levels. This integrated approach allows for the development of both preventive measures and reactive strategies, ensuring a comprehensive and effective response to drought.

While awareness of drought existed previously in the canton, it has taken concerted effort and numerous initiatives to achieve widespread institutional recognition and integrate drought into both cantonal risk analysis and civil protection efforts. For cantons initiating their cantonal risk analysis, the case study from Canton of Zurich demonstrates that while challenging, institutionalizing drought management is achievable through collaboration, strategic planning, and a commitment to proactive risk reduction.



Action Point 4 – Translate drought risk to local action

Drought as a complex hazard requires coordinated management across multiple levels of government. To effectively address this challenge, Action Point 4 focuses on translating identified drought risks to the local level, where the impacts on people, communities, ecosystems, and infrastructure are most acutely felt.

Using the Canton of Bern as a case study, this chapter explores strategies to foster collaboration between cantonal and local authorities. Expert interviews with cantonal civil protection officials highlighted that such collaboration is not always seamless, with challenges such as the lack of centralized information on water restrictions implemented by municipalities. To bridge this gap, the Canton of Basel-Landschaft launched the online platform Trockenheit⁹⁹ in the summer of 2024. This resource provides users with direct access to current drought regulations implemented at both cantonal and municipal levels. This initiative directly addresses a key finding in the 2024 Capability Analysis for Civil Protection,¹⁰⁰ which identified insufficient capabilities in data and situation management of water regulations (Trockenheitsfähigkeiten 104).

In the Canton of Bern, the initial response to an incident is handled by the relevant administrative division, in accordance with the principle of federalism. The cantonal Office for Civil Protection develops the cantonal hazard analysis, which includes scenario analysis, risk matrices, and recommendations. The municipalities are responsible for carrying out the required measures, such as creating emergency plans, under the oversight of the cantonal Office for Civil Protection to ensure a coordinated and efficient response.¹⁰¹

The updated cantonal hazard analysis, scheduled for release in 2025, will include drought as one of 25 hazards, influenced by the recommendations from the FOCP in the latest version of the National Risk Analysis.¹⁰² In contrast to other cantonal hazard analyses, the Canton of Bern categorizes drought as a distinct hazard and not in association with heatwaves or wildfires. This inclusion reflects a growing awareness of the canton's limited experience with prolonged hazards, such as drought, in comparison to hazards with short-term exposure, including landslides and floods. Moreover, the recent occurrence of droughts in the canton has brought the issue of the hazard to the attention of the general public and variIn the Canton of Bern, the cantonal civil protection authorities provide a detailed drought risk scale for each municipality, empowering local governments to develop targeted responses and prioritize preparedness efforts.

The online platform *Trockenheit*,⁹⁹ launched by the Canton Basel-Landschaft, serves to enhance transparency and accessibility of drought regulations across different administrative levels.

ous sectors of society. This has led to a greater awareness of the necessity for public education and outreach campaigns, as outlined in Action Point 9 "Conduct community education and outreach programs".

In the cantonal hazard analysis, the frequency of drought events is estimated to occur once every 10 to 100 years, affecting all the municipalities within the canton. The extent of a drought event is estimated to range from a relatively small occurrence of less than two million Swiss francs to a more significant event (with a projected cost of between 10-50 million Swiss francs) for municipalities of greater size. Furthermore, the cantonal hazard analysis also offers a specific drought risk assessment for each municipality. In the context of drought, these assessments concentrate on critical action priorities, such as "Unsecured water supply" and "Excessive groundwater abstraction." To ensure local acceptance and implementation, these risk assessments are approved by the respective municipal authorities, fostering trust and collaboration between authorities.

The case study of the Canton of Bern illustrates how the translation of risk from the national to the cantonal and local levels is necessary for local governments and communities to develop effective coping mechanisms for drought events.¹⁰³ An overview of this process is depicted in **Figure 6**.

In the Canton of Bern, the cantonal civil protection authorities disseminate risk information through the provision of a detailed drought risk scale for each municipality. This ensures that local communities are equipped with the necessary knowledge to understand the specific threats they face. This locally tailored information empowers and enables local authorities to develop targeted responses, such as water conservation measures, and to prioritize preparedness efforts, such as the drought exercise involving relevant local actors in Emmental, as outlined in Action Point 7 "Strengthen drought preparedness through exercises". Moreover, it facilitates collaboration between municipalities with comparable risk profiles. For instance, in order to mitigate potential future vulnerabilities and ensure a reliable water supply, municipalities in the Canton of Bern are actively engaged in creating networks to secure water availability for smaller communities. This practice is being increasingly adopted in other cantons as well.104

Furthermore, the Canton of Bern proactively involves local actors in the cantonal hazard analysis process, fostering a sense of ownership and accountability, which are pivotal elements for effective disaster risk reduction. This engagement is exemplified by the stipulation that municipal authorities approve their respective municipal risk assessment included in the cantonal haz-

Figure 6 illustrates the process of translating drought risk to local action



ard analysis. While the onus of developing emergency plans remains with the municipalities, this collaborative approach ensures that the process is not perceived as a top-down imposition but rather as a shared responsibility. It further demonstrates how robust accountability, manifested in the form of vertical steering by cantonal actors, can act as a bridge for coordination,¹⁰⁵ thereby enhancing the overall effectiveness of drought management efforts in the canton. Only an identified and communicated risk will allow for appropriate drought planning and management.

In conclusion, the case study of the Canton of Bern demonstrates the critical importance of translating drought risk from national to cantonal and local levels, enabling effective local response and preparedness. By providing detailed, municipality-specific drought risk assessments and involving local actors in the hazard analysis process, the canton fosters collaboration, accountability, and targeted action at the community level.



Action Point 5 – Develop a drought response plan

A well-structured drought response plan is an indispensable tool for any region preparing to address water scarcity. The plan serves as a dynamic framework that not only prepares for but also effectively responds to drought events. As a living document, it should be subject to periodic reviews and updates to ensure its continued relevance and effectiveness in the face of evolving conditions.

The goal of Action Point 5 is to demonstrate the necessity of a drought response plan and to offer assistance in preparing such a plan, with references to case studies from the cantons of Fribourg and Valais. A drought response plan should be developed in advance, based on thorough studies of the characteristics, impacts, and measures to alleviate drought while considering the social, political, and economic conditions of the area. Preparing the plan ahead of time allows for public feedback and input, ensuring that the responses align with the varying severity of water shortages.¹⁰⁶

At its core, a drought response plan provides a structured framework to address and manage drought conditions effectively. Such a framework accomplishes several critical objectives.¹⁰⁷ First, it equips the cantonal crisis management body with the necessary tasks within the civil protection mandate and allocates resources to coordinate and oversee the drought response efforts. This includes the establishment of clear lines of communication, the assembling of technical expertise, and securing adequate funding to implement response measures.

Second, the plan defines the roles and responsibilities of the stakeholders involved, ensuring that all parties are aware of their respective obligations, from government agencies and public service providers to individual citizens and businesses.

Third, the plan outlines a comprehensive communication strategy so that authorities and the public are kept informed about the evolving drought situation, the potential impacts, and the measures being taken to mitigate them. This bolsters public trust and cooperation during a crisis.

Fourth, the plan calls for collaboration and coordination among all levels of government, from the cantonal to the municipal level, as well as public service providers such as water utilities and power services. This directs different efforts to a unified response that leverages the strengths and resources of all involved parties. Figure 7 outlines the key elements that should be included in a drought response plan, referencing the plans of the Canton of Fribourg and the Canton of Valais.

Drought response plan

V Foundamental information

Includes clear objectives, scope of the plan, relevant legal framework, and precise definitions of drought and heatwave to establish a common understanding and guide decision-making.

Activation thresholds

Defines specific meteorological forecasts, such as temperature thresholds or precipitation deficits, that trigger different phases of the plan (e.g., pre-alert, alert, crisis) to ensure timely and appropriate action.

🗹 Risk assessment

Identifies potential impacts of drought and heatwaves across various sectors, including health (heatstroke, dehydration), infrastructure (e.g., water supply, agriculture), and the environment (e.g., wildfires).

V Role definition

Clearly outlines the roles and responsibilities of all actors involved in drought response, from government agencies to community organizations, to ensure coordinated and effective action.

V Procedures

Provides detailed step-by-step procedures for decision-making, communication, coordination, and resource allocation during each phase of the plan, facilitating a streamlined and efficient response.

V Preparedness measures

Outlines proactive measures to mitigate the impact of drought, such as public awareness campaigns on water conservation, early warning systems, and pre-positioning of resources.

Fifth, the plan establishes strategies to contain and minimize the adverse effects of drought on various sectors and communities. This may involve water restrictions, conservation measures, and financial assistance programs to support those most impacted by the drought. In Switzerland, the Cantons of Fribourg and Valais have each developed a drought contingency plan in response to the canton's hazard analysis and recent experiences with heat and drought. These plans, established by the cantonal crisis management bodies, share key elements to effectively address the challenges of drought, including activation thresholds, risk assessment, and role definition.

Sixth, the plan identifies and allocates the necessary resources, including personnel and material, to enable effective management of a drought. This could include the hiring of additional staff for the implementation of drought response activities, the procurement of equipment for the purpose of water conservation or distribution, and the stockpiling of emergency supplies.

Finally, the plan investigates and implements preventive measures to reduce a region's vulnerability to future droughts. Such measures may include the promotion of water conservation practices, the investment in drought-resistant infrastructure, or the diversification of water sources.

In Switzerland, the Cantons of Fribourg and Valais have each developed a drought response plan that is established by the cantonal crisis management body. These plans exhibit a number of key elements that collectively enable effective responses to the challenges of drought, as shown in **Figure 7.**

The development of a drought response plan in the Canton of Fribourg was prompted by the canton's hazard analysis (with the next update expected in 2025) and recent experiences with heat and drought events. Led by the cantonal crisis management body, the process fostered collaboration and integration between previously siloed departments and experts. A key achievement was the establishment of a competencies and responsibility matrix, which clarified specific actions, decisionmaking competency levels, and government involvement.

The plan institutionalizes the tasks of each involved cantonal office, including the highest executive authority in the canton, namely the Cantonal Council. It establishes a clear link between each action and the specific responsibility. Furthermore, the plan formalizes the tools utilized in response to drought, including the establishment of situational awareness with the support of soil moisture measuring stations. This, in turn, further contributes to an inter-cantonal and regional monitoring network. Additionally, the canton is currently in the process of establishing a Risk Observatory within the General Secretariat of the Security and Justice Department, with the objective of addressing drought and other risks.

To date, no event has triggered the implementation of the full plan. Nevertheless, certain elements have been initiated, including water transport in alpine regions and the implementation of response activities, predominantly to address the elevated risks associated with wildfires and heatwaves. The lessons learned from these experiences, in conjunction with the insights gained from drought exercises as outlined in Action Point 7 "Strengthen drought preparedness through exercises", could be used to refine and optimize the action plan in the future.

An important outcome of interactions with cantonal civil protection authorities is the realization that the most effective cantonal drought management strategies will not resolve water conflicts if municipalities are not inclusively involved, as described in Action Point 4 "Translate drought risk to local action". This is particularly pertinent in instances where there is a necessity to curtail water usage during periods of drought, with the management of water being a key task at the municipal level. Therefore, the support from the cantonal level is always to be delivered in a subsidiary manner. Respecting political competencies and acting level-appropriately is key to ensuring that the federal structure in Switzerland prevails and needs to be considered when designing organizational structures and action plans.

Moreover, to further enhance drought management, it is advisable to inform and potentially consult non-governmental actors, such as the Swiss Farmers' Association (*Schweizer Bauernverband*), to cultivate a broader understanding and support for drought mitigation efforts.

In conclusion, a well-structured drought response plan is not merely a static document but a dynamic tool for safeguarding a canton's water security. The case studies of Fribourg and Valais illustrate that such plans must be comprehensive, clearly define roles and responsibilities, and be adaptable to evolving conditions. A successful plan involves collaboration across sectors and authorities, institutionalizes key tasks, and employs tools like monitoring networks and risk observatories. Importantly, it recognizes the critical role of municipalities and non-governmental actors in ensuring the plan's operational effectiveness.



Action Point 6 – Appoint a drought task force

A well-defined drought response is the foundation for building drought resilience. Another critical component is the establishment of a dedicated drought task force responsible for coordinating a comprehensive response to drought conditions. This task force can bring together various stakeholders to ensure efficient management of resources and effective implementation of strategies to mitigate the impact of drought.¹⁰⁸ It is recommended that the task force not only operates during emergencies but also monitors conditions and promptly activates or advises on the activation of the response plan.

As evidenced by the survey results in Section 2.1, establishing a drought task force is a pivotal postdrought adaptation measure that has been widely adopted in many cantons after experiencing drought events. As of August 2024, 16 cantons have appointed a drought expert or task force. Action Point 6 emphasizes, therefore, the importance of establishing a drought task force, providing guidance on its roles and responsibilities, and highlighting its relevance through the case study of the Canton of Thurgau.

In order to effectively address the complex and multifaceted nature of drought and its impacts, the task force should comprise representatives from relevant departments and offices. Additionally, to expand the expertise and range of perspectives represented in the task force, members from the academic community (i.e., climatologists, policy specialists, planners), private sector actors, environmental organizations, and public interest groups should be included where possible, either as full members or as contributors to specialized working groups within the cantonal hazard analysis. It is to be expected that the composition of the task force will vary from one canton to another, mirroring each region's social, political, and economic landscape.

The responsibilities of the drought task force extend beyond the immediate response to an event. As new research needs or institutional gaps emerge, the task force should compile a list of the issues in question and make recommendations to relevant agencies and legislatures on how to address them.¹⁰⁹ The degree of public visibility of the task force will also fluctuate in accordance with the recency and severity of drought events. Regardless of the level of public attention, the task force must include members with strong communiThe case study on the Canton of Thurgau underlines the value of a drought task force led by the cantonal Office for Civil Protection and Military. This task force serves as an operational tool and a center of expertise in drought management. Other effective models include cantonal committees where civil protection collaborates (the Canton of Bern), orientation meetings held during the summer period (the Canton of Fribourg), and annual dialogue platforms (the Cantons of Basel-Stadt and Basel-Landschaft).

cation skills to either lead or support public communication, ensuring transparency, trust-building, and community engagement,¹¹⁰ as outlined in Action Point 10 "Embrace collaboration".

Drought risk in the Canton of Thurgau has been an ongoing concern for over a decade, leading to the inclusion of drought in the cantonal hazard analysis since 2013. However, as early as 2006, the canton appointed a drought task force/specialist staff *(Fachstab Trockenheit)*. This task force is constituted by relevant governmental actors and is responsible for the regular and continuous assessment of the drought situation in the canton. The task force is managed and coordinated by the Office for Civil Protection and Military, and its members include cantonal authorities responsible for the environment, water quality, water construction and hydrology, forestry, hunting and fishery, agriculture, public health, legal affairs, police, communications, and building insurance.¹¹¹

In more detail, the task force enables the authorities to exchange information, coordinate actions, and present a unified front when proposing measures to the executive government of the canton. Although the task force itself is not vested with the authority to make decisions, it ensures that each office provides information, measurements, and data drawn from its respective areas of expertise and responsibility. The diverse information is then collected and presented to the cantonal executive power. This multidisciplinary approach permits a comprehensive understanding and management of drought risks, as well as ensuring a cohesive, centralized, and unified communication to convey measures to the public.

Several advantages are observed by the creation and existence of a drought task force at a cantonal level, as emerged during the expert interviews with the cantonal authorities. One of the primary benefits of a cantonal task force is its capacity to harness and leverage local specificity and expertise. The impact of drought conditions can vary significantly across regions due to diverse environmental, economic, and social factors. By including representatives from various offices in the task force, it allows for the incorporation of local knowledge and tailors strategies to address the unique characteristics of each region. In the Canton of Thurgau, for instance, the task force has recognized the importance of considering geographic and ecological patterns, such as watershed boundaries, soil types, and vegetation cover for drought management rather than relying solely on political boundaries such as cantonal or municipal borders.

Another key advantage lies in the facilitation of inter-cantonal collaboration. The Canton of Thurgau's task force, for example, actively collaborates with neighboring cantons like St. Gallen and Zurich, with the objective of addressing shared challenges and developing solutions that are mutually beneficial.

The task force provides further value as an operational tool for drought response by serving as a platform where subject matter experts can collaborate, exchange knowledge, and translate their expertise into practical applications. This institutionalizes know-how and functions as a cohesive, whole-of-government body, legitimizing its decisions and proposals to the public.

In cases where a formal drought task force does not already exist or there are limited resources to create one, civil protection authorities leverage alternative approaches to address the issue. For example, the Canton of Bern has set up a drought working group within the cantonal Office for Water and Waste. Bern's civil protection authorities can collaborate with this group to voice their perspectives and address challenges related to drought.

Another effective format has been observed in several cantons, whereby an orientation meeting is convened before or during the summer, where drought is discussed holistically with other hazards, such as heatwaves and wildfires. The Canton of Fribourg has adopted this approach, while the Canton of Basel-Stadt has expanded the concept to establish an annual dialogue platform with its neighboring Canton of Basel-Landschaft. This dialogue addresses concerns related to drought, wildfires, and fishery in the region.

At the federal level, the forthcoming crisis organization of the Federal Administration foresees having a dedicated specialist staff (*Fachstab Trockenheit*), as mentioned during the expert interviews outlined in Section 2.2. This development marks a notable political shift to distinguish drought from the existing natural hazard specialist staff (*Fachstab Naturgefahren*). This step recognizes the complexities of droughts that require a dedicated working group similar to other prolonged crises, such as migration, pandemics, or energy security.

The creation of a national drought specialist staff further improves the Federal Council's capacity for crisis anticipation, also marked by the Capability Analysis for Civil Protection, and will position it better to deploy a cross-departmental crisis organization in a timely manner.¹¹² It is yet to be defined how the drought specialist staff, intended to be activated only during crises, will operate and interface with other permanent Standing Core Units, and what role civil protection organizations will play at both the cantonal and federal levels.

In conclusion, Action Point 6 recommends the establishment of a drought task force or working group as a fundamental step in managing drought. This is corroborated by the multitude of cantons that formed such task forces following their own experiences with drought events. Whether implemented at the cantonal or federal level, these task forces serve as indispensable expert groups, advising on adequate measures and acting as mediators to resolve water conflicts.



Action Point 7 – Strengthen drought preparedness through exercises

One persistent issue in emergency response is the discrepancy between knowledge and action. Despite awareness of effective practices in emergency response, individuals and organizations may not necessarily translate this knowledge into behavioral change.^{113, 114} One effective method for addressing this challenge is the use of simulation exercises among crisis and emergency management organizations striving to close the gap between learning and action. This approach links informal and formal knowledge through action and experience.¹¹⁵ The diverse exercise programs for civil protection, as shown in **Figure 8**, prompt decision-makers, planners, and operational personnel to anticipate the potential progression of a drought and identify proactive measures to mitigate its impacts or cope with its effects.

Action Point 7 of the Drought Mitigation Toolkit sheds light on the importance of drought-related exercises in fostering organizational learning and addressing the learning-action gap. Exercises provide an invaluable instrument for assessing the strengths and weaknesses of existing structures and processes related to drought management, as well as for identifying potential avenues for improvement. Moreover, they serve as a venue for developing essential individual and collaborative skills, ultimately contributing to institutional knowledge.¹¹⁶ By actively engaging in exercises, civil protection organizations can effectively close the capability gap identified in Field of Action 3 of the Capability Analysis for Civil Protection that calls for joint exercises in civil protection (*Handlungsfeld 3: Verbundübungen im Bevölkerungsschutz*).¹¹⁷

The most well-developed drought management plans can be undermined if there is a lack of experience in implementing them. To this end, civil protection organizations should avail themselves of simulation exercises as a means to acquire the requisite experiences before an actual drought event. Disaster simulation exercises offer a controlled test environment to evaluate organizational arrangements and procedures.¹¹⁸ They may involve senior management at the strategic level but also personnel on the operational and tactical levels of the organizations. The value of conducting these exercises is considered high among crisis practitioners, as they provide arenas for learning and knowledge sharing.¹¹⁹ They help to internalize drought management plans, as outlined in Action Point 5 "Develop a drought response

Although exercises for drought within Swiss civil protection organizations are still rare, notable examples are paving the way. At the regional level, the exercise conducted by the regional crisis management body in Langnau in the Canton of Bern demonstrated the successful testing of civil protection responses to a drought scenario. The exercise Odescalchi 2022, with its cross-border component, yielded institutional benefits beyond immediate water shortage management. Additionally, the tabletop exercise "Drought" seeks to bring together representatives from cantonal and federal civil protection organizations to enhance preparedness for future extreme events.

plan", into organizational practices. Moreover, exercises deepen relationships and cooperation among participants (municipalities, cantons, or even countries), emphasizing the importance of Action Point 10 "Embrace collaboration".

In the domain of crisis and emergency management, operation-based exercises, including drills, functional exercises, and full-scale exercises, are regularly employed for the purpose of training response teams and testing the coordination and interoperability of resources and personnel.¹²⁰ In Switzerland, operation-based exercises related to drought are rare, as reported by civil protection authorities in question 15 of the survey in Section 2. This is despite the fact that civil protection authorities expressed interest in such exercises during the expert interviews. If exercises are conducted, they often involve only one civil protection partner organization within a single canton. One example is given by the Canton of Graubünden where personnel from the civil Defense exercise the operation of new heavy equipment for drinking water supply during an emergency scenario, as reported during the expert engagement process.

In other instances, these exercises are limited to a specific region, such as the exercise carried out by the regional crisis management body (regionales Führungsorgane, RFO) Langnau in the Canton of Bern, which tested the preparedness of local civil protection organizations for an extreme drought scenario in Emmental.¹²¹ In this example, the involvement of different civil protection organizations, even though tested at a local scale, proved successful in responding holistically to a drought scenario impacting several sectors of the region.

Large-scale exercises involving different civil protection partner organizations and multiple cantons targeting drought remain relatively uncommon, mainly due to their inherent complexity and resource demands. Larger exercises also tend to be logistically demanding, requiring collaboration and deployment of personnel

Figure 8 provides an overview of exercise programs applicable to civil protection and emergency management

Graph inspired from Emergency Management Program from Massachusetts Institute of Technology, and Emergency Management Institute of the US Federal Emergency Management Agency-FEMA

Continuum of Exercises



Operations-Based

from various institutional organizations. In light of their considerable expense and time-consuming nature, these exercises are reserved for addressing the most critical hazards and functions. Nevertheless, a noteworthy example of a drought-related full-scale exercise occurred in 2022 in the Canton of Ticino within the realm of "Odescalchi 2022."122 This exercise entailed a significant mobilization of personnel and resources, reinforcing cross-border collaboration and mutual aid. This was achieved through the implementation of a new memorandum of understanding between the Cantons of Ticino, Valais, Graubünden, and the Italian border provinces.¹²³ Odescalchi 2022 included civil protection partner organizations from the Canton of Ticino and military personnel working in cooperation with their Italian counterparts to refine skills and mutual understanding in military and civil assistance in the event of a disaster. The first of the eight scenarios tested during the exercise addressed a drinking water shortage in Mendrisiotto (the southernmost district of the Canton of Ticino and Switzerland), where a prolonged drought prevented the authorities from supplying the population district with drinking water. The simulation exposed challenges in the distribution aqueduct, prompting industrial companies in Mendrisio to seek aid from the cantonal civil protection authorities. In response, civil protection partner organizations, in collaboration with the military, intervened by collecting water from the nearby Lake Ceresio, transporting it to Mendrisio, and purifying it for consumption via the deployment of mobile water filtration treatment plants. These mobile water treatment plants, provided by the Canton of Basel-Stadt, are frequently deployed to address drinking water shortages during emergencies such as water shortage, water pollution, or industrial accidents. The sharing of assets occurred through the Federal Resource Management Mechanism (*Ressourcenmanagement Bund-ResMaB*), reiterating the importance of collaboration and resource sharing during crises.

The main aspects of the exercise were linked to technical and operational methods for responding to emergencies. However, the exercise remains a successful attempt to test institutional and operational drought preparedness. It allowed for the identification of areas for improvement in the domain of crisis response, testing of new assets, such as mobile water filtration treatment plants, and different forms of collaboration. This includes cross-cantonal collaboration for the purpose of resource sharing, international collaboration to refine mutual support, and civil-military collaboration to enhance mutual understanding of crises and emergencies.

An alternative to operation-based exercises is discussion-based exercises, which include seminars, workshops, and tabletop exercises.¹²⁴ Specifically, tabletop exercises entail the convening of relevant stakeholders to deliberate and rehearse their assigned duties in the face of a hypothetical scenario, with facilitators guiding the discussion. Tabletop exercises are often employed to familiarize the participants with plans, policies, and procedures or even to discuss or develop new ones.¹²⁵ Tabletop exercises have proven to be a helpful method to address complex emergencies and to improve communication and information flow, as well as to provide a venue for conflict resolution between the involved organizations.126

Although the literature on tabletop exercises targeting drought is limited, notable exceptions exist, primarily in the United States. For example, the South Carolina Drought Tabletop exercise, developed by the South Carolina Emergency Operations Center in 2019, gathered participants from national emergency response teams to local water managers and Federal Agencies to learn about roles and responsibilities in drought response and to exercise drought preparedness and response actions.¹²⁷

In Switzerland, a tabletop exercise titled "Drought"¹²⁸ is scheduled for November 20, 2024. This exercise is designed to evaluate the readiness of the Swiss civil protection system in managing multi-year drought events. During the event, members of civil protection partner organizations will be exposed to hypothetical yet realistic extreme drought scenarios developed by the FOCP, the WSL Extremes program, and the Center for Security Studies at ETH Zurich. This approach allows for a comprehensive examination of existing structures, processes, and resources at the cantonal and federal levels, encompassing both operational and strategic components to enhance preparedness for future extreme events.

This tabletop exercise aims to challenge participants to think more comprehensively and clearly about the consequences of drought and the responses to it. By bringing together different domains of Swiss civil protection organizations, the exercise seeks to further raise awareness of drought exposure and generate innovative ideas for improving drought planning and response strategies before the next drought occurs.

In conclusion, bridging the gap between knowledge and action is essential for enhancing drought preparedness within Switzerland's civil protection system. Whether operation-based or discussion-based, simulation exercises provide invaluable platforms for testing response strategies, refining procedures, and fostering collaboration among stakeholders. These exercises not only build experience and confidence in implementing drought management plans but also highlight areas for improvement. As drought hazards continue to grow in significance, expanding the scope and frequency of such exercises will be crucial in strengthening the resilience of civil protection organizations and ensuring they are wellequipped to manage future challenges.



Action Point 8 – Participate in knowledge exchange

One of the striking findings from the survey of cantonal civil protection authorities in Section 2 was their expressed interest in participating in a national knowledge exchange program focused on drought management and resilience. In light of this interest, Action Point 8 explores the creation of a national knowledge exchange platform and its prospective benefits for improving drought resilience in Switzerland. This Action Point also postulates the essential role of incorporating academic expertise into the platform, ensuring scientific advice for drought management.

A national platform can strengthen and unify the efforts of the various stakeholders engaged in drought management and resilience. By functioning as a central nexus for practitioners, decision-makers, academics, and other relevant parties, it consolidates diverse perspectives and expertise.¹²⁹ Given the varying perceptions of drought and its impacts on individual interests, experiences, and contexts, the platform's initial objective is to integrate these multiple forms of knowledge and perspectives on drought.¹³⁰ This integration is intended to create a repository of innovative ideas and applications, facilitate joint exercises, as outlined in Action Point 7 "Strengthen drought preparedness through exercises", and overcome potential barriers to knowledge generation and resilience development that may arise from differing understandings of drought risk.

In addition to sharing knowledge, the platform would promote networking initiatives to facilitate discussions and the exchange of experiences. This aspect is of particular importance, as the expert interviews with cantonal authorities have revealed a lack of awareness regarding drought management measures implemented in other cantons. Active participation in the platform's networks would enable cantons to gain a broader understanding of drought challenges and solutions that extend beyond their individual borders, thereby improving their capacity to respond to drought events. The existing platform Conference of Cantonal Authorities for the Military, Civil Protection, and Civil Defense (Konferenz der kantonalen Verantwortlichen für Militär, Bevölkerungsschutz und Zivilschutz)131 demonstrates the effectiveness of inter-cantonal collaborative discussions within the civil protection domain.

The knowledge exchange program could be integrated into existing forums or established as a standCantonal civil protection authorities' expressed interest in knowledge sharing highlights the need for a platform that facilitates discussions, joint exercises, and innovative solutions. This corroborates informed and effective drought response strategies. Action Point 8 highlights the importance of a national knowledge exchange platform for drought management, unifying efforts and integrating scientific expertise with practical experience to enhance resilience across Switzerland.

alone entity. Regardless of its format, the platform's success hinges on the active participation of scientific actors whose research and expertise would complement the practical knowledge of practitioners and decision-makers. By incorporating scientific insights into every phase of the integrated risk management cycle,¹³² from planning and event documentation to post-drought evaluation, the platform would foster informed decision-making and effective response to future drought challenges.¹³³

This collaborative approach, where drought researchers and planners work together at the interface of science and policy, aligns with the core goals and methodology of this report and mirrors the Federal Council's vision for the upcoming dedicated drought specialist staff (*Fachstab Trockenheit*) outlined in Action Point 6 "Appoint a drought task force". The resulting synergy between theory and practice would ensure that drought management strategies are evidence-based and grounded in real-world experience, maximizing their effectiveness and impact.

One of the principal areas where science and civil protection practitioners intersect is in the documentation process during and after drought events. In Switzerland, the Federal Government is expected to introduce a cantonal reporting obligation in the event of drought conditions.¹³⁴ This reporting would elaborate on observed water deficits and conflicts and protective measures for ecosystems. They would also present an analysis of the potential implications of future drought events, including the number of exceptional authorizations granted for emergency water use. However, an independent study conducted in 2022 and commissioned by FOEN revealed that 19 cantons do not have an established procedure for systematically collecting information on water uses and protection during drought periods.135 The application of scientific knowledge can address this gap by developing standardized protocols and methodologies for the collection and analysis of data, ensuring consistency and comparability across cantons. Such data, when integrated with post-drought evaluations, can inform evidencebased decision-making and support the implementation of targeted interventions.

The undertaking of post-drought evaluations is fundamental for civil protection. Such evaluations serve to preserve institutional memory and facilitate the process of learning from past successes and mistakes. These evaluations should analyze the climatic and environmental aspects of the drought, its economic and social consequences, and the effectiveness of pre-drought planning, relief efforts, and recovery initiatives.¹³⁶ Crucially, they should identify both the weaknesses and the strengths in drought-coping mechanisms, highlighting areas where societies exhibited resilience and not just failures. To support an unbiased assessment, the responsibility for postdrought evaluation and societal responses can be delegated to independent institutions, such as universities and specialized research institutes.¹³⁷

Although civil protection organizations are not directly involved, examples of comprehensive postdrought evaluation reports from the Cantons of Aargau¹³⁸ and Zurich¹³⁹ demonstrate the value of detailed insights and recommendations for future improvements in drought management.

In conclusion, the establishment of a national knowledge exchange platform is endorsed by cantonal civil protection authorities and represents a promising avenue for enhancing Switzerland's drought resilience. This collaborative forum, combined with proactive measures outlined in previous Action Points, has the potential to markedly enhance Switzerland's resilience to drought and its associated impacts and consequences.



Action Point 9 – Conduct community education and outreach programs

Droughts impact communities in multifaceted ways based on their severity, duration, and spatial extent.¹⁴⁰ The goal of Action Point 9 is to build drought resilience through consistent education and outreach programs. It is important to recognize that resilience does not imply immunity to drought, but rather, it signifies the capacity to mitigate its impact through early action. Raising awareness and managing knowledge about drought is essential for reducing its risks. Civil protection organizations, which have traditionally been geared towards immediate disaster response, need to expand their role to address slow-onset, complex, and prolonged crises like drought. This requires a modification in communication strategies, particularly when attempting to disseminate information regarding the intricacies of droughts in contrast to short-term weather events like heavy rainfall and flash floods.141

A significant increase in public awareness of drought has been observed in Switzerland, as evidenced by the expert interviews in Section 2.2. This awareness has been noted across the general public and various sectors and is largely attributable to recent drought events as well as ongoing efforts and initiatives to combat climate change. Despite this awareness, past drought events have shown that cantonal civil protection authorities still face challenges in navigating the complex discussions with communities about the rationalization and prioritization of water use, especially in the event of disputes.142,143 While Switzerland does not currently face an acute water scarcity crisis, brief periods of water shortage have highlighted how prioritization of water use remains a critical issue. For instance, in emergencies, including prolonged droughts, cantons may adjust the minimum residual flow rates to allow for temporary water withdrawals, especially for drinking water supplies, as stipulated in Art. 32 (d) of the Water Protection Act.144 The law, however, does not specify the relative priorities between using water for cooling and other industrial applications, firefighting, or agricultural irrigation, leading to conflicts in some municipalities. The Capability Analysis for Civil Protection identifies conflicts over water arising from rationing or prioritization issues as a gap in civil protection capabilities in Switzerland's civil protection capabilities for managing droughts (Trockenheitsfähigkeiten 103).145

Engaging with the public ensures acceptance and compliance with water restriction measures, which are crucial for effective drought management. Action Point 9 emphasizes that public drought information, such as the cantonal drought bulletins in St. Gallen and Graubünden, and engagement activities, like the walking tours in Basel-Landschaft, are instrumental in building drought resilience. These initiatives equip communities with the knowledge and tools needed to face drought with preparedness and adaptability rather than crisis.

Disputes arise not only in relation to the allocation of water among different stakeholders but also in instances where restrictions are imposed in one administrative division but not in the neighboring ones.^{146, 147}

Additionally, response strategies implemented by civil protection authorities or the cantonal and regional crisis management bodies, often with the support of other actors, can be contentious. A recurring challenge involves the deployment of the federally anchored capacity of the armed forces to transport water to sustain alpine communities. This practice is intended for emergencies and should only occur if the cantons' assets and resources, including the ones of the private sector, are exhausted and the criteria of subsidiarity are fulfilled. However, this intervention has become more frequent in the past. For instance, at the end of summer 2022, the armed forces transported 585,000 liters of water via helicopter to alpine farms.¹⁴⁸

To address these disputes and conflicts, yearround community engagement and outreach campaigns from cantonal and federal authorities are necessary to raise awareness about the various impacts of drought and dispel common misconceptions surrounding the issue.

Public outreach campaigns often focus narrowly on providing information on how to reduce water usage.¹⁴⁹ For example, communication measures during the summer drought of 2022 were largely limited mainly to appeal to the population by individual municipalities or water suppliers to reduce drinking water consumption.¹⁵⁰ However, effective community engagement and educational programs must address a broader range of information. Comprehensive educational programs should inform the public about the sources of their water, who uses it, the historical and potential future impacts of drought on their community, and the benefits of proactive drought planning.¹⁵¹ Greater understanding is likely to lead to greater compliance when citizens are requested to comply with water consumption restrictions in response to drought. The survey results in Section 2.1 revealed that cantonal civil protection authorities encounter challenges in engaging the public in water conservation

efforts. When individuals are involved in the decisionmaking process or are provided with clear and transparent information about the necessity of restrictions, they are more likely to perceive the measures as fair and necessary, reducing resistance and promoting voluntary cooperation.¹⁵² Moreover, it is also likely that individuals will begin to place more value on the long-term steps the community can take to be better prepared for the next drought.

Furthermore, engaging with the community helps to better capture and compare the variety of perceptions on drought phenomena and to gather insights from various stakeholders to understand diverse perspectives on drought and its impacts.¹⁵³ Because drought is challenging to recognize until it manifests through location- and sector-specific impacts,154 different actors understand it in different ways. Capturing what drought means for different individuals, organizations, and sectors and mapping how this understanding evolves, including if stakeholders are prepared for it, are key components of risk management to reduce drought impacts. For instance, when developing the national monitoring and EWS outlined in Action Point 1 "Establish and incorporate a national drought early warning system", the responsible federal offices initially conducted surveys and targeted interviews with a variety of users across different cantons. These endeavors tailored and shaped the development of the system based on the real needs of users.

Proactive communication and information sharing are essential in keeping the public informed on an evolving drought situation. For example, the Cantons of St. Gallen¹⁵⁵ or Graubünden¹⁵⁶ provide regular updates on the current drought situation through bulletins referencing parameters such as temperature, precipitation, surface water runoff, and groundwater level. This transparency helps to build trust and fosters a shared understanding of the challenges posed by drought, encouraging greater community participation in conservation efforts.

Beyond public communication, proactive engagement also extends to key stakeholders. For instance, the Canton of Thurgau offers a detailed checklist for water suppliers,¹⁵⁷ guiding their preparation for and response to increased water demands during droughts, particularly from the agricultural sector. This targeted approach ensures that water resources are managed effectively and equitably during periods of scarcity.

Finally, community outreach through excursions or guided walks to drought-affected areas and water infrastructure sites (e.g., reservoirs or treatment plants) can be an effective method for the co-production of knowledge and public education.¹⁵⁸ The underlying premise is that these experiences evoke personal memories and shared narratives by establishing connections between people and places, enriching knowledge exchange and collaborative problem-solving. These activities promote informal dialogue and collaboration among a diverse range of stakeholders, including scientists, policymakers, farmers, and community members. For example, the excursion in the Riehen-Bettingen forest organized by Canton of Basel-Landschaft¹⁵⁹ is a primary initiative that offers a unique opportunity to discuss and explore solutions for water efficiency and conservation, promoting public social norms that value water as a precious resource.

In conclusion, fostering open dialogue, public awareness, and education empowers communities to not only withstand the effects of drought but also to mitigate its impact. Action Point 9 demonstrates that public engagement and outreach are key to building resilience to drought and paving the way for a future where drought is met not with crisis but with preparedness and adaptability.



Action Point 10 – Embrace collaboration

Droughts transcend political borders, often escalating from local to national and even international levels. However, the regulations and resources available to address them – such as time, personnel, and assets – are limited and confined within geographical boundaries. While water scarcity can lead to conflicts across different sectors, organizations, and communities, Action Point 10 champions collaboration as an effective means to mitigate drought risk across society. By breaking down silos, facilitating information exchange, and pooling resources, collaboration enables a more coordinated and effective response across sectors and communities.

A collaborative approach extends across all levels of government, civil society, and the private sector. It transcends mere cooperation or resource sharing and involves the concerted efforts of diverse entities working together to overcome organizational boundaries, levels of authority, and sectoral divisions.¹⁶⁰ The overarching objective is to collectively prepare for, respond to, and recover from drought events.

The necessity for collaboration in the context of drought management is evident at various levels. The findings of the survey and the expert interviews with cantonal civil protection authorities reveal a lack of crossborder collaboration, limiting engagement between Swiss cantons as well as with neighboring European countries. In fact, the cross-border management of shared water resources often leads to controversy and tensions, underscoring the need for collaborative solutions.^{161, 162} This need also extends to the local level. In the Canton of Lucerne, the absence of a collaborative approach has resulted in fragmented water supply systems and a lack of cantonal oversight,¹⁶³ an issue mirrored in other cantons as well. Additionally, recent drought events and water conflicts in the Canton of Aargau have emphasized the importance of regional cooperation in water supply planning, highlighting the necessity for cross-sectoral collaboration.164

Rather than a constraint, collaboration should be viewed as an enabling mechanism, as it offers a multitude of benefits. First, it saves costs by pooling resources and avoiding duplication of efforts. Second, the long-term protection of people and nature is becoming increasingly complex, and the burden of addressing these challenges alone is overwhelming for individual entities. Third, effective solutions require a collective approach that goes beDrought transcends political boundaries, necessitating collaborative solutions across all levels of government, civil society, and the private sector to effectively prepare for, respond to, and recover from drought events. Action Point 10 emphasizes the need for collaboration within the civil protection system as a strategic necessity for mitigating the impacts of drought, pooling resources, and fostering a shared responsibility for water resource management.

yond the interests of individual stakeholder groups, ensuring that all perspectives are represented and incorporated into the planning and response procedures. This inclusive approach can foster a sense of shared responsibility and ownership over water resources, leading to more sustainable and equitable outcomes.

Moreover, collaboration with academic institutions - through what is often referred to as transdisciplinary approaches – ensures that societal and scientific stakeholders can co-create knowledge.¹⁶⁵ This means that research, for example, supports drought risk assessments and provides valuable insights into possible future scenarios, taking into account local conditions provided by local experts (e.g., emergency managers for natural hazards). This, in turn, helps the relevant authorities (e.g., civil protection) to define appropriate crisis management plans for effective preparation and response, as well as for recovery capacities. An example of this process, is the Research Program Extremes, which fosters inter- and transdisciplinary research to equip Swiss stakeholders with appropriate decision-making tools and coping strategies to address future extremes, including drought.¹⁶⁶

While the need for collaboration is evident and there is a strong desire for it across all levels, as shown by the willingness of most survey respondents to engage in both inter-cantonal and international cooperation, initiating and maintaining effective partnerships can be challenging. Diverse organizational structures, hierarchical levels, and different responsibilities create a complex web of demands that can be hard to navigate. However, several initiatives are underway to overcome these challenges.

The Canton of Basel-Landschaft's new Water Strategy, for example, provides a model that specifies the responsibilities of each level and office of the cantonal government in implementing water measures.¹⁶⁷ The model is intended as a guide for the canton and its municipalities to build trust (through transparent communication, shared goals, and mutual respect), define roles and responsibilities (to prevent overlaps and ensure accountability), develop joint protocols, and invest in capacity building. Additionally, a new initiative consisting of workshops brings together researchers, members of the Federal Administration, including FOCP, and the insurance industry to explore synergies in drought management, fostering a multi-disciplinary approach to drought risk assessment and mitigation.¹⁶⁸ Another example of a permeable government collaboration is given by the Canton of Bern. Its risk assessments for the municipalities described in Action Point 4 "Translate drought risk to local action" were calculated based on data and statistics from several organizations other than civil protection.

Furthermore, Switzerland actively participates in international collaborations on disaster management. This includes engagement within the UNDRR, the UN's central coordinating body for disaster risk reduction, and active membership in the High-level Meeting on National Drought Policy (HMNDP) held in Geneva. The HMNDP encourages national governments to adopt policies that foster cooperation and coordination at all levels to enhance their capacity to cope with drought-induced water scarcity.¹⁶⁹

On a more localized level, the International Lake Constance Conference serves as a central hub for crossborder cooperation between Swiss cantons and the neighboring regions of Germany, Austria, and Lichtenstein, facilitating dialogue and joint action among stakeholders, decision-makers, and practitioners, including civil protection authorities.¹⁷⁰

For cantonal civil protection authorities, collaboration can take the form of joint exercises to simulate drought scenarios and test response mechanisms, as described in Action Point 7 "Strengthen drought preparedness through exercises". It can also involve the sharing of assets, the exchange of knowledge and expertise through different existing platforms, or the establishment of new dedicated forums, as illustrated in Action Point 8 "Participate in knowledge exchange".

In conclusion, droughts often occur during periods of elevated demand for water, thereby testing the capacity of individual communities to respond effectively. The deployment of limited resources, including time, personnel, and specialized equipment, can be rapidly exceeded in the event of a drought. Action Point 10 concludes the presentation of the Drought Toolkit, emphasizing that from a civil protection perspective, collaboration is not just a desirable option but a strategic necessity for mitigating the impacts of drought.

4 Concluding Remarks

Switzerland is often referred to as Europe's water tower, but it is not immune to the growing risk of drought. Droughts are extreme weather events with far-reaching and often devastating consequences for people, infrastructure, the economy, and ecosystems. Unlike many other natural hazards, their effects unfold gradually, impacting diverse areas such as agriculture, human health, energy security, and public safety. They disrupt the hydrological cycle and ecosystem functions, materializing over weeks, months, or even years. However, this gradual onset offers a window of opportunity for early intervention and the development of adaptation strategies.

Civil protection organizations are key to building resilience to such slow-onset crises. Their mandate to protect the population and its livelihoods extends beyond immediate emergencies and encompasses longterm risk reduction. The Drought Mitigation Toolkit presented in this report provides a roadmap for transitioning the Swiss civil protection system from emergency management to risk reduction in the context of drought. Grounded in research and expert insights, this toolkit offers 10 Action Points that form a framework for developing regional drought management policies and preparedness plans. It provides guidance rather than rigid prescriptions, encouraging reflection and discussion even if full implementation is not immediately feasible. After all, fostering dialogue about drought's relevance to civil protection is a core objective of this report. While tailored to the Swiss context, the toolkit's methodology can be adapted to various levels of government worldwide.

The diverse case studies highlighted in the toolkit reveal numerous ongoing drought mitigation initiatives across Switzerland. However, these efforts often remain isolated within their respective institutions or geographic areas. This report aims to connect these disparate initiatives by showcasing them and providing contact points for knowledge exchange and collaboration. The fragmented nature of current drought management practices further underscores the need for a national drought plan, one that harmonizes efforts, facilitates information sharing, and ensures a cohesive response across cantons or local administrations. If implemented, Switzerland would follow suit with other European countries that have national drought plans or are in the process of developing them.¹⁷¹

Though this report primarily focuses on strategic and structural aspects of drought management within the Swiss civil protection system, it is important to acknowledge that political decisions regarding water allocation and resource management will ultimately determine the success of these efforts. Addressing the complex challenges of drought requires not only technical solutions or operational actions but also a willingness to make difficult choices and prioritize long-term sustainability over short-term gains.

Looking ahead, drought will continue to be a pressing concern for Switzerland and the journey towards drought resilience is ongoing. This report serves as a call to encourage discussions and exchanges among civil protection authorities, decision-makers, and communities in Switzerland regarding their vulnerability to drought and its potential impact on people, infrastructure, sectors, ecosystems, and institutional systems.

5 Bibliography

- 1 Kamberaj J. et al., "Trend Analysis Civil Protection 2035: Uncertainties, Challenges and Opportunities", *Risk and Resilience Report*, Center for Security Studies (CSS), ETH Zürich, 02.2024.
- 2 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, 26.06.2024, <u>newsd.admin.ch</u>.
- 3 Bundeskanzlei, Krisenmanagement, <u>bk.admin.ch</u>.
- 4 Bundesamt für Bevölkerungsschutz BABS, *Inte*grales Risikomanagement, 2019, <u>babs.admin.ch</u>.
- 5 Bundesamt für Bevölkerungsschutz BABS, *Leitfaden Kataplan – die kantonale Gefährdungsanalyse*, <u>babs.admin.ch</u>.
- 6 Bundesamt für Bevölkerungsschutz BABS, «Gefährdungsdossier Trockenheit», *Katastrophen und Notlagen Schweiz*, 2020, <u>babs.admin.ch</u>.
- 7 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, 26.06.2024, <u>newsd.admin.ch</u>.
- 8 Federal Office of Meteorology and Climatology MeteoSwiss, *Drought*, <u>meteoswiss.admin.ch</u>.
- 9 Federal Office of Meteorology and Climatology MeteoSwiss, *Drought*, <u>meteoswiss.admin.ch</u>.
- 10 National Drought Mitigation Center, *Types of Drought*, <u>drought.unl.edu</u>.
- 11 Duel H. et al. "HELP Guiding Principles for Drought Risk Management under a Changing Climate", *High-level Experts and Leaders Panel on Water and Disasters (HELP)*, 2022. <u>cms.deltares.nl</u>.
- 12 Berg A. and Sheffield J., "Climate Change and Drought: The Soil Moisture Perspective." *Current Climate Change Reports* 4, no. 3, pp. 180–91, 2018, SpringerLink.
- 13 Stubbington R. et al., "The Effects of Drought on Biodiversity in UK River Ecosystems: Drying Rivers in a Wet Country", *WIREs Water* 11, no. 5, 2024, Wiley Interdisciplinary Reviews.

- 14 World Meteorological Organization and Global Water Partnership, "National Drought Management Policy Guidelines: A Template for Action (D.A. Wilhite)", Integrated Drought Management Programme (IDMP) Tools and Guidelines Series 1, 2014, droughtmanagement.info.
- 15 World Meteorological Organizations, *Drought,* <u>wmo.int</u>.
- 16 National Centers for Environmental Information, Billion-Dollar Weather and Climate Disasters-United States Summary, <u>ncei.noaa.gov</u>.
- 17 Federal Department of Foreign Affairs FDFA, *About Switzerland: Lakes and Rivers*, <u>eda.admin.ch</u>.
- 18 Federal Office of Meteorology and Climatology MeteoSwiss, *Drought indices*, <u>meteoswiss.admin.ch</u>.
- 19 Federal Office of Meteorology and Climatology MeteoSwiss, *Climate change*, <u>meteoswiss.admin.ch</u>.
- 20 Kittl B., Stroheker S. and Blaser S., "Slight Increase in Infestations with Spruce Bark Beetle in 2022", Swiss Federal Research Institute WSL, 7.03.2023. wsl.ch.
- 21 Büchli Ch. «Kanton Aargau Besorgt Wasser Wird Knapper» in: *SRF*, 10.07.2023. <u>srf.ch</u>.
- 22 Ibrahim S., "Water Shortage: Switzerland's Blue Gold is Under Pressure", in: *SWI swissinfo*, 12.06.2023, <u>swissinfo.ch</u>.
- 23 Weise Z., "When the Water Runs Dry: Why France is Freaking Out Over a Tiny Swiss Dam", in: *Politico*, 10.05.2024, <u>politico.eu</u>.
- Jorio L., "How to Prevent Conflicts Over Water in the Middle of Europe", in: *SWI swissinfo*, 26.06.2023. <u>swissinfo.ch</u>.
- 25 Copernicus, *OBSERVER: 2022: a year of extremes,* 29.12.2022, <u>copernicus.eu</u>.
- 26 Imfeld N. et al., «Ein Ziemlich Normaler Zukünftiger Sommer», *Geographica Bernensia*, Geographisches Institut, University of Bern, 2022.
- 27 World Economic Forum, *Droughts are creating new supply chain problems. This is what you need to know*, 25.10.2023, <u>weforum.org</u>.

- 28 European Environmental Agency, *Drought impact* on ecosystems in Europe, 10.10.2023, <u>eea.europa.eu</u>.
- 29 Tratschin R. et al., «Trockenheit im Sommer 2022», EBP im Auftrag des Bundesamtes für Umwelt BAFU, 24.12.2023. <u>bafu.admin.ch</u>.
- 30 Jezella C. «Hunderttausende Liter Wasser Verschwunden», in: *Berner Oberländer*, 20.07.2023. <u>berneroberlaender.ch</u>.
- 31 Imfeld N. et al., «Ein Ziemlich Normaler Zukünftiger Sommer», *Geographica Bernensia*, Geographisches Institut, University of Bern, 2022.
- 32 Die Bundesversammlung der Schweizerischen Eidgenossenschaft, *SR 520.1 Bundesgesetz über den Bevölkerungsschutz und den Zivilschutz*, Art. 2, 01.01.2022, <u>fedlex.admin.ch</u>.
- 33 Die Bundesversammlung der Schweizerischen Eidgenossenschaft, *SR 520.1 Bundesgesetz über den Bevölkerungsschutz und den Zivilschutz*, Art. 2, 01.01.2022, <u>fedlex.admin.ch</u>.
- 34 Bundesamt für Bevölkerungsschutz BABS, *Leitfaden KATAPLAN. Grundlage für kantonale Gefährungsanalysen und Massnahmenplanungen*, 2013, <u>babs.admin.ch</u>.
- 35 Staatskanzlei Kanton Schaffhausen, *Teilrevision des Bevölkerungsschutzgesetzes und des Zivilschutzgesetzes; Eröffnung der Vernehmlassung*, 30.11.2023, <u>sh.ch</u>.
- Regierungsrat Kanton Bern, Zeitgemässe Regelungen für Bevölkerungsschutz und Zivilschutz,
 24.03.2023, <u>be.ch</u>.
- 37 Regierungsrat Kanton Thurgau, *Bevölkerungs*schutzgesetz soll angepasst werden, 3.05.2024, tg.ch.
- 38 Regierungsrat Kanton Basel-Stadt, *Bericht zur Corona-Bewältigung im Kanton Basel-Stadt liegt vor, 20.03.2024, <u>regierungsrat.bs.ch</u>.*
- 39 Canton du Valais-Service de la sécurité civile et militaire, *Protection de la population – Gestion des situations d'urgence*, 15.03.2023, <u>vs.ch</u>.
- 40 Kanton Aargau, Gesetz über den Bevölkerungsschutz und den Zivilschutz im Kanton Aargau (Bevölkerungsschutz- und Zivilschutzgesetz Aargau, BZG-AG); Änderung, <u>ag.ch</u>.

- 41 Der Bundesrat, *Bundesrat eröffnet Vernehmlassung zur Verordnung über die Krisenorganisation*, 15.05.2024. <u>admin.ch</u>.
- 42 Bundesamt für Bevölkerungsschutz BABS, «Verordnung über die Krisenorganisation der Bundesverwaltung (KOBV)», Erläuternder Bericht zur Eröffnung des Vernehmlassungsverfahrens, 2025, newsd.admin.ch.
- 43 Aldunce P. et al., "Local Perception of Drought Impacts in a Changing Climate: The Mega-Drought in Central Chile", *Sustainability* 9, no. 11, 2017, MDPI.
- Raikes J. et al., "Pre-disaster Planning and Preparedness for Floods and Droughts: A Systematic Review." International Journal of Disaster Risk Reduction 38, 2019, Elsevier.
- 45 Raikes, J., et al. "Pre-disaster Planning and Preparedness for Floods and Droughts: A Systematic Review." International Journal of Disaster Risk Reduction 38, 2019, Elsevier.
- 46 Global Water Partnership Central and Eastern Europe. Guidelines for Preparation of the Drought Management Plans, p. 31, 2015, <u>gwp.org</u>.
- 47 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, p. 27, 26.06.2024, <u>newsd.admin.ch</u>.
- 48 United Nations Office for Disaster Risk Reduction, United Nations General Assembly, *Report of the open-ended intergovernmental expert working group on indicators and terminology relating to disaster risk reduction*, 1.12.2016, <u>undrr.org</u>.
- 49 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, p. 27, 26.06.2024, <u>newsd.admin.ch</u>.
- 50 Bundesamt für Bevölkerungsschutz BABS, *Inte*grales Risikomanagement, 2019, <u>babs.admin.ch</u>.
- 51 Raikes J. et al., "Pre-disaster Planning and Preparedness for Floods and Droughts: A Systematic Review." International Journal of Disaster Risk Reduction 38, 2019, Elsevier.
- 52 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, p. 27, 26.06.2024, <u>newsd.admin.ch</u>.

- 53 Logar I. and van den Bergh J., "Methods to Assess Costs of Drought Damages and Policies for Drought Mitigation and Adaptation: Review and Recommendations", *Water Resources Management* 27, no.6, pp. 1707–1720, 2013.
- 54 Kanton Zürich-Kantonale Führungsorganisation, «Risikomanagement Bevölkerungsschutz 2021», <u>zh.ch</u>.
- 55 Daschkeit A., «Klimawandel Einfluss auf den Bevölkerungsschutz», *Crisis Prevention – Fachportal für Gefahrenabwehr, Innere Sicherheit und Katastrophenhilfe,* 5.06.2019.
- 56 Kamberaj J. et al., "Trend Analysis Civil Protection 2035: Uncertainties, Challenges and Opportunities", *Risk and Resilience Report*, p. 13, Center for Security Studies (CSS), ETH Zürich, 02.2024.
- 57 Eriksen C. et al., "Adapting to Climate Change: Lessons for Swiss Civil Protection", *Risk and Resilience Report*, p. 29, Center for Security Studies (CSS), ETH Zürich, 07.2023.
- 58 EBP Schweiz AG und Bundesamt für Bevölkerungsschutz, Auswirkungen des Klimawandels auf den Bevölkerungsschutz in der Schweiz, 2021, ncss.admin.ch.
- 59 Prior T. and Roth F., "Learning from Disaster Events and Exercises in Civil Protection Organizations", *Risk and Resilience Report*, Center for Security Studies (CSS), 2016. ETH Zurich.
- 60 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, p. 27, 26.06.2024, <u>newsd.admin.ch</u>.
- 61 Regierungskonferenz Militär, Zivilschutz, und Feuerwehr, *Konferenz der kantonalen Verantwortlichen für Militär, Bevölkerungsschutz und Zivilschutz,* <u>rkmzf.ch</u>.
- 62 Bundesamt für Bevölkerungsschutz BABS, *Inte*grales Risikomanagement, 2019, <u>babs.admin.ch</u>.
- 63 World Meteorological Organization and Global Water Partnership, "National Drought Management Policy Guidelines: A Template for Action (D.A. Wilhite)", *Integrated Drought Management Programme (IDMP) Tools and Guidelines Series 1*, 2014, <u>droughtmanagement.info</u>.

- 64 Sustainable Development Goals, *Drought Toolbox Launched at UNCCD COP 14*, 17.09.2019, <u>sdg.iisd.org</u>.
- 65 FAO, Drought portal-Knowledge resources on integrated drought management, <u>fao.org</u>.
- 66 United Nations, *Early Warnings for All*, <u>un.org</u>.
- 67 Zappa M. et al., "A Prototype Platform for Water Resources Monitoring and Early Recognition of Critical Droughts in Switzerland", *Evolving Water Resources Systems: Understanding, Predicting and Managing Water – Society Interactions,* Proceedings of ICWRS2014, 2014.
- 68 Lenkungsausschuss Intervention Naturgefahren LAINAT, *Optimierung der Warnung und Alarmierung OWARNA: zweiter Folgebericht*, 2018, <u>newsd.</u> <u>admin.ch</u>.
- 69 Bundesamt für Bevölkerungsschutz BABS, «Gefährdungsdossier Trockenheit», *Katastrophen und Notlagen Schweiz*, 2020, <u>babs.admin.ch</u>.
- 70 Twigg J., "Chapter 16: Preparing for Disasters and Emergencies", *Disaster Risk Reduction Mitigation and Preparedness in Development and Emergency Programming*, Humanitarian Practice Network, 2004.
- 71 Der Bundesrat, *Trockenheit: Bundesrat will nationales System zur Früherkennung und Warnung*, 18.05.2022, <u>admin.ch</u>.
- 72 Der Bundesrat, *Trockenheit: Bundesrat will nationales System zur Früherkennung und Warnung*, 18.05.2022, <u>admin.ch</u>.
- 73 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, p. 32, 26.06.2024, <u>newsd.admin.ch</u>.
- 74 Lenkungsausschuss Intervention Naturgefahren LAINAT, *Optimierung der Warnung und Alarmierung OWARNA: zweiter Folgebericht*, 2018, <u>newsd.</u> <u>admin.ch</u>.
- Popovic N., Asseburg J. and Weber S., "Communicating Weather Warnings to the Swiss Population

 Insights of a Representative Online Study."
 Scientific Report MeteoSwiss 106, 2023.

- Wilhite, D., "Integrated Drought Management: Moving from Managing Disasters to Managing Risk in the Mediterranean Region", *Euro-Mediterranean Journal for Environmental Integration*, pp. 4–42, 2019. SpringerLink.
- 77 National Integrated Drought Information System, Drought Assessment in a Changing Climate: Priority Actions & Research Needs, 29.11.2023, drought.gov.
- 78 Schreier S., «Wegen Trockenheit und Hitze: Könnte der Rhein in Basel Austrocknen?», In: *BZ Basel*, 15.07.2022, <u>bzbasel.ch</u>.
- 79 Schweizerische Rheinhäfen, *Jahresbericht 2023,* port-of-switzerland.ch.
- 80 Schweizerische Rheinhäfen, *Jahresbericht 2023,* port-of-switzerland.ch.
- 81 Muccione V. et al., «Klimawandel Schweiz: Auswirkungen Kombinierter Klimarisiken auf Urbane Systeme.» *Factsheet zur Fallstudie «Kombinierte Klimarisiken» in Urbanen Systemen,* EClim Research Group on Environment and Climate, <u>eclim-re-</u> <u>search.ch</u>, 2022.
- 82 Bundesamt für Umwelt BAFU: Hydrologische Daten und Vorhersagen, *Rhein – Weil, Palmrainbrücke*, <u>hydrodaten.admin.ch</u>.
- 83 Kanton Aargau, *Umwelt-Ereignisdokumentation Trockenheit Jahre 2018 und 2019*, <u>ag.ch</u>.
- 84 Neubert M., «Zu Wenig oder zu Warmes Wasser in Rhein und Aare: Droht den Atomkraftwerken die Überhitzung?», In: *Südkurier,* 16.07.2022, <u>suedkurier.de</u>.
- 85 Muccione V. et al., «Klimawandel Schweiz: Auswirkungen Kombinierter Klimarisiken auf Urbane Systeme.» *Factsheet zur Fallstudie «Kombinierte Klimarisiken» in Urbanen Systemen,* EClim Research Group on Environment and Climate, <u>eclim-re-</u> <u>search.ch</u>, 2022.
- 86 Schweizer Radio und Fernsehen, *Badeverbot in der Birs und der Wiese*, 12.08.2020, <u>srf.ch</u>.
- 87 Bundesamt für Bevolkerungschutz BABS, *Bericht zur nationalen Risikoanalyse Katastrophen und Notlagen Schweiz 2020*, 1.11.2020, <u>babs.admin.ch</u>.

- Wilhite, D., "Integrated Drought Management: Moving from Managing Disasters to Managing Risk in the Mediterranean Region", *Euro-Mediterranean Journal for Environmental Integration*, pp. 4–42, 2019, SpringerLink.
- 89 Regierungsrat des Kantons Basel-Stadt, *Bericht zur Corona-Bewältigung im Kanton Basel-Stadt liegt vor*, 20.03.2024, <u>regierungsrat.bs.ch</u>.
- 90 European Commission-Joint Research Center, *Cross-border and emerging risks in Europe*, 2024, jrc. <u>eu.europa.eu</u>.
- 91 World Bank. Assessing Drought Hazard and Risk: Principles and Implementation Guidance, p. 10, 2019, gfdrr.org.
- 92 Das Bundesamt für Bevölkerungsschutz BABS, Katastrophe und Notlagen Schweiz 2020, Methode zur nationalen Risikoanalyse, 2020, <u>babs.admin.ch</u>.
- 93 Kanton Zürich, *Bevölkerungsschutzgesetz (BSG)*, 01.07.2008, <u>zh.ch</u>.
- 94 Bundesamt für Bevölkerungschutz BABS, *Leitfaden KATAPLAN-Kantonale Gefährdungsanalyse und Vorsorge*, 2013, <u>babs.admin.ch</u>.
- 95 Kanton Zürich, *Bevölkerungsschutzgesetz (BSG)*, 01.07.2008, <u>zh.ch</u>.
- 96 Kanton Zürich-Kantonale Führungsorganisation, «Risikomanagement Bevölkerungsschutz 2021», <u>zh.ch</u>.
- 97 Federal Office for the Environment FOEN, *Pilot Programme 'Adaptation to climate change'*, 03.09.2018, <u>admin.ch</u>.
- 98 EBP Schweiz AG und Bundesamt für Bevölkerungsschutz, Auswirkungen des Klimawandels auf den Bevölkerungsschutz in der Schweiz, 2021, ncss.admin.ch.
- 99 Kanton Basel-Landschaft, <u>Trockenheit</u>: *Massnahmen und Bestimmungen in Bezug auf Trockenheit von Kanton und Gemeinden*, <u>baselland.ch</u>.
- 100 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, p. 37, 26.06.2024, <u>newsd.admin.ch</u>.

- 101 Kanton Bern, *Kantonales Bevölkerungsschutz- und Zivilschutzgesetz (KBZG)*, Art. 3, <u>belex.sites.be.ch</u>.
- 102 Bundesamt für Bevölkerungsschutz BABS, «Gefährdungsdossier Trockenheit», *Katastrophen und Notlagen Schweiz*, 2020, <u>babs.admin.ch</u>.
- 103 International Strategy for Disaster Reduction and World Bank, Drought Risk Reduction Framework and Practices: Contributing to the Implementation of the Hyogo Framework for Action, 2007, <u>unisdr.org</u>.
- 104 BärnToday. «Nur Wenige Berner Gemeinden Sorgen Sich Wegen Wassermangel»,10.08.2022, <u>baerntoday.ch</u>.
- 105 Lieberherr E. and Ingold K., "Actors in Water Governance: Barriers and Bridges for Coordination", *Water* 11, no. 2, 2019, MDPI.
- 106 Nazif S., Tavakolifar H., Eslamian S., "Emergency Drought Consequence Plan." In: *Handbook of Drought and Water Scarcity*, pp. 641–659. CRC Press, 2017.
- 107 World Meteorological Organization and Global Water Partnership, "National Drought Management Policy Guidelines: A Template for Action (D.A. Wilhite)", Integrated Drought Management Programme (IDMP) Tools and Guidelines Series 1, 2014, droughtmanagement.info.
- 108 World Meteorological Organization and Global Water Partnership, "National Drought Management Policy Guidelines: A Template for Action (D.A. Wilhite)", Integrated Drought Management Programme (IDMP) Tools and Guidelines Series 1, 2014, droughtmanagement.info.
- 109 Wilhite D. et al., "The Basics of Drought Planning: A 10-Step Process", *National Integrated Drought Information System*, 2000.
- 110 Wilhite D. et al., "The Basics of Drought Planning: A 10-Step Process", *National Integrated Drought Information System*, 2000.
- 111 Kanton Thurgau, *Kanton Thurgau setzt Fachstab Trockenheit ein,* 4.07.2023, <u>tg.ch</u>.
- 112 Der Bundesrat, *Bundesrat verbessert Organisation des Krisenmanagements*, 29.03.2023, <u>admin.ch</u>.

- 113 Müller A. et al., "Emergency drills for agricultural drought response: a case study in Guatemala", *Disasters*, 43:2, pp. 140–430, 2019, Wiley.
- 114 Hill H. et al., "The Invitational Drought Tournament: What Is It and Why Is It a Useful Tool for Drought Preparedness and Adaptation?" *Weather and Climate Extremes* 3, pp. 107–116, 2014, Science Direct.
- 115 Stern E., "Designing Crisis Management Training and Exercises for Strategic Leaders". *CRISMART* Volume 42. National Defense College.
- 116 Prior T. and Roth F., "Learning from Disaster Events and Exercises in Civil Protection Organizations", *Risk and Resilience Report*, Center for Security Studies (CSS), 2016. ETH Zurich.
- 117 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, p. 22, 26.06.2024, <u>newsd.admin.ch</u>.
- 118 Prior T. and Roth F., "Learning from Disaster Events and Exercises in Civil Protection Organizations", *Risk and Resilience Report*, Center for Security Studies (CSS), 2016. ETH Zurich.
- Grunnan T. and Fridheim H., "Planning and Conducting Crisis Management Exercises for Decision-Making: The Do's and Don'ts." *EURO Journal on Decision Processes* 5, pp. 79–95, 2017, Elsevier.
- 120 World Health Organization, *WHO Simulation Exercise Manual*, 2017, <u>iris.who.int</u>.
- 121 Langnau im Emmental Gemeinde, ErnstfallÜbung «Extreme Trockenheit, 13.05.2023, langnau-ie.ch.
- 122 Republica e Cantone Ticino. *Odescalchi 2022: molti scenari di crisi e la firma di un nuovo Protocollo d'intesa tra Ticino e Italia,* 17.05.2022, <u>ti.ch</u>.
- 123 Kellerhais H., «ODESCALCHI 22: Alleine Viel, Zusammen Viel Mehr», In: *Allgemeine Schweizerische Militärzeitschrift*, <u>asmz.ch</u>.
- 124 Federal Emergency Management Agency, *Types of Training and Exercises*, emilms.fema.gov.

- 125 Grunnan T. and Fridheim H., "Planning and Conducting Crisis Management Exercises for Decision-Making: The Do's and Don'ts", *EURO Journal on Decision Processes* 5, pp. 79–95, 2017, Elsevier.
- 126 Thornton M. et al., "The Effectiveness of Table-Top Exercises in Improving Pandemic Crisis Preparedness", International Information and Engineering Technology Association, pp. 463–471, 2021.
- 127 Lackstrom K., *South Carolina Drought Table Top Exercise-Final Report*, 2019, <u>cisa.sc.edu</u>.
- 128 Swiss Federal Institute for Forest, Snow, and Landscape Research WSL, *Tabletop «Drought»: Wie bewältigt der Schweizer Bevölkerungsschutz zukünftige Trockenheitsereignisse?*, <u>wsl.ch</u>.
- 129 Kamberaj J. et al., "Trend Analysis Civil Protection 2035: Uncertainties, Challenges and Opportunities", p. 48, *Risk and Resilience Report*, Center for Security Studies (CSS), ETH Zürich, 02.2024.
- Grainger S., Murphy C., Vincente-Serrano S.,
 "Barriers and Opportunities for Actionable Knowledge Production in Drought Risk Management: Embracing the Frontiers of Co-production", Sec. Interdisciplinary Climate Studies 9, 2022, Frontiers.
- 131 Regierungskonferenz Militär, Zivilschutz, und Feuerwehr, Konferenz der kantonalen Verantwortlichen für Militär, Bevölkerungsschutz und Zivilschutz, rkmzf.ch.
- 132 Bundesamt für Bevölkerungsschutz BABS, *Inte*grales Risikomanagement, 2019, <u>babs.admin.ch</u>.
- 133 Wilhite, D., Sivakumar M., Pulwarty R., "Managing Drought Risk in a Changing Climate: The Role of National Drought Policy", Weather and Climate Extremes 3, 2014, ScienceDirect.
- 134 Der Bundesrat, «Wasserversorgungssicherheit und Wassermanagement», *Grundlagenbericht Bericht des Bundesrates in Erfüllung des Postulates 18.3610 Rieder vom 15. Juni 2018*, 12.12.2021, <u>newsd.admin.ch</u>.
- 135 Tratschin R. et al., «Trockenheit im Sommer 2022», *EBP im Auftrag des Bundesamtes für Umwelt BAFU*,
 24.12.2023. <u>bafu.admin.ch</u>.

- 136 Wilhite D., "Essential Elements of National Drought Policy: Moving Toward Creating Drought Policy Guidelines", *Drought Mitigation Center Faculty Publications 79*, 2011. <u>core.ac.uk</u>.
- 137 Wilhite, D., Sivakumar M., Pulwarty R., "Managing Drought Risk in a Changing Climate: The Role of National Drought Policy", *Weather and Climate Extremes 3*, 2014, ScienceDirect.
- 138 Canton Aargau-Departement Bau, Verkehr und Umwelt, *Ereignisdokumentation Trockenheit, Jahre* 2018 und 2019, 2020, <u>ag.ch</u>.
- 139 Kanton Zürich-Baudirektion, Züricher Umweltpraxis-Schwerpunk Lehren aus dem Trockensommer 2018, <u>zh.ch</u>.
- 140 Clifford K. et al., "Rapidly Assessing Social Characteristics of Drought Preparedness and Decision Making: A Guide for Practitioners", *Social Science Methods*, U.S. Department of the Interior and U.S. Geological Survey, 2022, <u>pubs.usgs.gov</u>.
- 141 Twigg J., "Chapter 16: Preparing for Disasters and Emergencies," In *Disaster Risk Reduction Mitigation and Preparedness in Development and Emergency Programming*, Humanitarian Practice Network, 2004.
- 142 Gall Ch., «Wem gehört das Wasser? Konflikte werden zunehmen», In: *SRF*, 30.07.2022, <u>srf.ch</u>.
- 143 Büchli Ch. «Kanton Aargau Besorgt Wasser Wird Knapper»in: *SRF*, 10.07.2023. <u>srf.ch</u>.
- 144 Der Bundesrat, *Bundesgesetzüber den Schutz der Gewässer vom 24. Januar 1991 (Stand am 1. Februar 2023),* <u>fedlex.data.admin.ch</u>.
- 145 Der Bundesrat, «Fähigkeitsanalyse Bevölkerungsschutz», *Bericht Des Bundesrates*, p. 37, 26.06.2024, <u>newsd.admin.ch</u>.
- 146 Wunderlin T., «Zürcher Bauern Pumpen Wasser aus demselben Bach, aus dem die Wasserentnahme im Thurgau Verboten ist», In: *Thurgauer Zeitung*, 4.08.2022, <u>thurgauerzeitung.ch</u>.
- 147 ZüriToday, «Zürcher Bauern nutzen Wasser aus Bach – Thurgauer dürfen nicht», 28.07.2022, <u>zueritoday.ch</u>.

- 148 Akademie der Naturwissenschaften Schweiz, *SRF Einstein, Wasserknappheit in der Schweiz*, 2022, <u>scnat.ch</u>.
- 149 California Drought Preparedness, *Public Outreach*, <u>cadroughtprep.net</u>.
- 150 Tratschin R. et al., «Trockenheit im Sommer 2022»,
 EBP im Auftrag des Bundesamtes für Umwelt BAFU,
 24.12.2023. <u>bafu.admin.ch</u>.
- 151 National Oceanic and Atmospheric Administration, National Drought Mitigation Center, the Oklahoma Climatological Survey, the Illinois State Water Survey and State Climatologist's Office, and the Lower Platte River Corridor Alliance, Drought-Ready Communities: A Guide to Community Drought Preparedness, 2011, drought.unl.edu.
- 152 Jorgensen B., Graymore M., O'Toole K., "Household Water Use Behavior: An Integrated Model", *Journal of Environmental Management 91*, no. 1, 2009, Elsevier.
- 153 Teutschbein C., "Drought Hazards and Stakeholder Perception: Unraveling the Interlinkages between Drought Severity, Perceived Impacts, Preparedness, and Management", *Ambio 52*, 2023, Springer-Link.
- 154 Poděbradská M., et al. "Ready for Drought? A
 Community Resilience Role-Playing Game", *Water*12, no. 9, 2020, MDPI.
- 155 Kanton St. Gallen, Trockenheit Bulletin, sg.ch.
- 156 Kanton Graubünden, Trockenheit Bulletin, gr.ch.
- 157 Canton Thurgau-Amt für Umwelt, *Checkliste Umgang mit Wasserknappheit/Trockenheit,* <u>umwelt.tg.ch</u>.
- 158 Grecksch K., "Waterways Walk: Eliciting Information from Stakeholders by Connecting People and Places", *SEE 2019 Conference Co-creation – Making Ecological Economics Matter*, 2019.
- 159 WaldSchweiz, Wald-Wasser-Trockenheit, Exkursion im Landschaftspark Wiese und im Wals von Riehen, 26.06.2024, <u>waldschweiz.ch</u>.
- 160 Kamberaj J. et al., "Trend Analysis Civil Protection 2035: Uncertainties, Challenges and Opportunities", *Risk and Resilience Report*, p. 23., Center for Security Studies (CSS), ETH Zürich, 02.2024.

- 161 Weise Z., "When the Water Runs Dry: Why France Is Freaking Out Over a Tiny Swiss Dam", In: *Politico*, 10,05.2024. <u>politico.eu</u>.
- Jorio L., "How to Prevent Conflicts Over Water in the Middle of Europe", In: *SWI swissinfo*, 26.06.2023. <u>swissinfo.ch</u>.
- 163 Kanton Luzern, *Wassernutzung und Wasserversorgung im Kanton Luzern: Aktueller Stand und künftige Herausforderungen,* p. 10, 2021, <u>uwe.lu.ch</u>.
- 164 Canton Aargau-Departement Bau, Verkehr und Umwelt, *Ereignisdokumentation Trockenheit, Jahre* 2018 und 2019, p. 17, 2020, <u>ag.ch</u>.
- 165 Pohl C. et al., "Conceptualising Transdisciplinary Integration as a Multidimensional Interactive Process", *Environmental Science & Policy* 118, pp. 18–26, 2021, ScienceDirect.
- 166 Swiss Federal Institute for Forest, Snow, and Landscape Research, WSL Research Programme Extremes 2021–2025, <u>wsl.ch</u>.
- 167 Kanton Basel-Landschaft, *Wasserstrategie Kanton Basel-Landschaft Gemeindeseminar*, 10.11.2023, <u>baselland.ch</u>.
- 168 Schweizer Hagel, *Workshop Future Extremes-Exploring synergies*, 30.05.2024, <u>LinkedIn</u>.
- 169 Wilhite, D., "Integrated Drought Management: Moving from Managing Disasters to Managing Risk in the Mediterranean Region", *Euro-Mediterranean Journal for Environmental Integration*, pp. 4–42, 2019, SpringerLink.
- 170 Kanton St.Gallen, *Internationale Bodensee-Konferenz (IBK)*, <u>sg.ch</u>.
- Schmidt G. et al., "Stock-Taking Analysis and Outlook of Drought Policies, Planning and Management in EU Member States", *Final Report*, p. 10, 2023. European Commission. <u>op.europa.eu</u>.



The **Center for Security Studies (CSS) at ETH Zürich** is a center of competence for Swiss and international security policy. It offers security policy expertise in research, teaching and consulting. The CSS promotes understanding of security policy challenges as a contribution to a more peaceful world. Its work is independent, practice-relevant, and based on a sound academic footing.